A STATISTICAL STUDY ON THE PREDICTION OF RANDOM NOISE FLUCTUATION AND THE MUTUAL RELATIONSHIP BETWEEN VARIOUS NOISE EVALUATION INDICES IN AN ACOUSTIC ENVIRONMENT

Yasuo Mitani

- · · · ·

January 1990

Dissertation submitted to the Faculty of the Systems Engineering Course of the Graduate School of Engineering of Hiroshima University in partial fulfillment of the requirements for the degree of Doctor of Engineering

#### ABSTRACT

Industrial growth carries with it an increasing environmental noise pollution problem. In order to evaluate quantitatively environmental random noise, many statistical noise descriptors (i.e., noise evaluation indices) are now being used. Percentile noise level Lx, equivalent noise level Leg, traffic noise index TNI and perceived noise level PNL are some examples. Almost all these statistical indices are obtained in relationship with the whole probability distribution of environmental random noise. Noise surveys have indicated that road traffic noise is a major component of environmental random noise. In this doctoral thesis, a statistical method for prediction of road traffic noise generated by actual traffic flow of an arbitrary non-Poisson type is first discussed for practical application. Paying attention to the fact that sound insulation barriers are very often constructed to produce attenuation of environmental random noise, a method for predicting the stochastic insulation effect of a sound insulation barrier is then discussed for the case of arbitrary probability distribution of random noise incident on the barrier. In addition, the mutual relationships between various types of statistical noise evaluation indices in relation to  $L_{X}$  and  $L_{eq}$ , which play an important role in the field of noise evaluation and regulation problems, are discussed for the purpose of evaluating systematically real random noise. More specifically, a method for estimating an  $L_{eq}$  evaluation index is proposed using the moment statistics of random noise fluctuation. Next, a method for estimation of  $L_{eq}$  is presented, using the known values of several specific  $L_x$ . A new trial estimating the original distribution form of random noise fluctuation is then presented using the known values of  $L_{eq}$  and several specific  $L_x$ , and a specific on-line measurement system based on the above-mentioned estimation theory for  $L_x$  and  $L_{eq}$  is constructed with the aid of a microcomputer by introducing an iterative process for extracting the moment statistics of random noise fluctuation. The theoretical approaches in this study are generally applicable to any kind of random phenomena. The theoretical results are experimentally confirmed by applying them to actually observed noise data.

### CONTENTS

INTRODUCTION .....

1

. . . . . . . .

# STATISTICAL METHOD FOR PREDICTION OF RANDOM NOISE FLUCTUATION

PART I

Chapter 1 A Method for Prediction of Road Traffic Noise Generated	
from Non-Poisson Type Traffic Flow	8
1.1 Introduction	8
1.2 Theoretical consideration	9
1.2.1 Description of problem	9
1.2.2 Arbitrary non-linear function type statistics	
for an arbitrary non-Poisson type traffic flow	11
1.2.3 Prediction of road traffic noise by use of	
moment statistics for the noise intensity	
fluctuation	15
1.3 Experimental consideration	21
1.3.1 Simplification of an elementary response wave	
form by use of triangular shape approximation	21
1.3.2 Predicted results based on the proposed method	22
1.4 Conclusion	28

Chapter 2 A Method for Prediction of Road Traffic Noise at	
a T-Type Road Intersection Based on the Image	
Method	31
2.1 Introduction	31
2.2 Theoretical consideration	31
2.2.1 Outline of prediction theory	31
2.2.2 Estimation of the elementary response wave form	
using the image method	34
2.3 Experimental consideration	44
2.4 Conclusion	47
Chapter 3 A Statistical Method for Prediction of the Sound	
Insulation Effect of Barriers	49
3.1 Introduction	49
3.2 Theoretical consideration	50
3.2.1 General expression for noise intensity	
distribution	50
3.2.2 Relationship between frequency characteristics	
of a barrier and parameters contained in the cumulative	
distribution function	52
3.2.3 Estimation of the frequency characteristic of a sound	
insulation barrier	53
3.3 Experimental consideration	55
3.4 Conclusion	56

## PART II

# MUTUAL RELATIONSHIP BETWEEN VARIOUS TYPES OF NOISE EVALUATION INDICES CONNECTED WITH $\rm L_x$ AND $\rm L_{eq}$

Chapter 1 A Method for Estimation of $L_{eq}$ Noise Evaluation	
Index by Use of Moment Statistics	60
1.1 Introduction	60
1.2 Theoretical consideration	62
1.2.1 General theory for estimation of mean value of	
arbitrary non-linear function type random	
variables	62
1.2.2 Estimation method of $L_{\text{eq}}$ noise evaluation index $\ldots\ldots$	65
1.3 Experimental consideration	68
1.4 Conclusion	74
Chapter 2 Mutual Relationship between Several Noise Evaluation	
Indices Connected with ${\rm L}_{\rm x}$ and ${\rm L}_{\rm eq}$	75
2.1 Introduction	75
2.2 Theoretical consideration	76
2.2.1 Method for estimation of $L_{eq}$ noise evaluation index	
by use of several specific ${\tt L}_{\tt X}$	76
2.2.2 Method for estimation of noise probability distribution	
by use of ${\rm L}_{\rm eq}$ and several specific ${\rm L}_{\rm x}$	79
2.3 Experimental consideration	83

	2.4	Conclusion	91
(	Chapter	3 Further Investigation of Method for Estimation of	
		Noise Probability Distribution by Use of $\mathbf{L}_{\text{eq}}$ and	
		Several Specific ${\rm L}_{\rm X}$	95
	3.1	Introduction	95
	3.2	Theoretical consideration	96
	3.3	Experimental consideration	101
	3.4	Conclusion	103
(	Chapter	$4$ Calculation of $\rm L_{x}$ and $\rm L_{eq}$ Noise Evaluation Indices	
		by Use of Moment Statistics and Their	
		Microcomputer-Aided On-Line Measurement	107
	4.1	Introduction	107
	4.2	Theoretical consideration	109
	4.2	2.1 General algorithm for estimating $L_{eq}$	109
	4.2	2.2 General algorithm for estimating ${\rm L}_{\rm X}$	111
	4.3	Experimental consideration	113
	4.3	3.1 Confirmation of effectiveness of the proposed	
		estimation method	113
	4.3	3.2 Construction of an on-line measurement system for	
		sequential measurement	118
	4.4	Conclusion	124

-

CONCLUSION .		25
--------------	--	----

ACKNOWLEDGMENTS	131	
REFERENCES	133	
LIST OF THE AUTHOR'S PUBLICATIONS	138	

INTRODUCTION

• •

#### Introduction

In recent years, problems resulting from environmental noise generated by various forms of transformation and industry have become more critical. In order to quantitatively assess real random noises, it is essential that a precise method for evaluation of these random noises taking human responses to these stochastic phenomena into consideration be found. When the problem of statistical evaluation of such stochastic phenomena is discussed, the following problems appear to be primary importance from the methodological viewpoint :

- (1) Many kinds of statistical noise evaluation indices, such as  $L_{eq}$ , TNI,  $L_{NP}$  and  $L_x$  (x=5, 10,...; i.e., (100-x) percentage point of noise level distribution) have been proposed, almost all of which can be obtained in close relation to the entire probability distribution.
- (2) Environmetal random noises encountered in daily life give various kinds of probability distribution form other than the well-known Gaussian distribution, owing to the diversity of causes of fluctuation.
- (3) Two of the many noise evaluation indices,  $L_X$  and  $L_{eq}$ , play an important role in the field of noise evaluation and regulation.
- (4) In order to evaluate environmental random noise accurately, methods of prediction other than the well-known conventional methods derived from simplified models or standard probability expressions must be employed because of various factors. Not only physical

-1-

factors but also other more complicated factors related to environmental conditions, individual psychologic response, etc., must be taken into account.

These practical points are dealt with in this doctoral thesis in a statistical method for prediction of noise level distribution and the mutual relationships between various kinds of evaluation indices in close relation to  $L_x$  and  $L_{eq}$ .

This thesis is divided into an Introduction, Parts I and II, consisting of Chapters 1-3 and Chapters 1-4 respectively the Conclusion and References. The outline of each part and chapter is given below.

In Part I, a general method for predicting the probability distribution form of environmental random noise is considered. In the problem of environmental noise control, a method for systematic prediction of the statistical properties of environmental noise is naturally of great importance. Furthermore, attention must be paid to the fact that environmental noise pollution is caused primarily by road traffic noise.

Chapter 1 discusses a statistical method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow. Until now, various approaches for predicting road traffic noise have been proposed from various points of view. These prediction methods can essentially be divided into the following two groups. One group comprises structural methods based on physical models. The other comprises functional methods based on mathematical models. The

-2-

former is constructed by accounting for various types of physical mechanisms such as traffic flow, propagation characteristics, etc. Thus, the more such physical mechanisms are taken into account in detail, the more complex the prediction method itself becomes. The latter is constructed by pre-establishing a mathematical model for the actual noise fluctuation itself. If only the latter method is used, quantitative inference of physical clues to practical noise control is not possible. Accordingly, a kind of hybrid method combining the latent advantages of the two prediction groups was proposed in a previous study. In order to establish a prediction method of this kind for road traffic noise generated from actual traffic flow of an arbitrary non-Poisson type, a new approach for practical use equivalent to the approach used for idealized Poisson type traffic flow is proposed in this chapter. The proposed method is applicable to treat any kind of road traffic noise fluctuation generated from actual non-Poisson type traffic flow, once information on the elementary response wave form of the level time pattern observed when one vehicle passes directly in front of an observation point has been obtained. The effectiveness of the proposed prediction method is experimentally confirmed by applying it to actually observed road traffic noise data.

Chapter 2 discusses a practical method for prediction of road traffic noise at a T-type road intersection based on the image method. In the previous chapter, a general method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow based on information on the elementary response wave form observed

-3-

experimentally is proposed. In order to establish a more effective method of prediction, a theoretical method for estimating the abovementioned elementary response wave form, especially in a complex realistic acoustical environment must be found. In this chapter, a new approach for theoretical estimation of this elementary response wave form is proposed for a T-type road intersection with sound insulation barriers, using a well-known image method. The effectiveness of the proposed method is confirmed experimentally by applying it to road traffic noise data obtained at a T-type road intersection in a city area.

Chapter 3 describes a study on the probabilistic response of a sound insulation barrier. In the practical engineering field of noise control, sound insulation barriers are often constructed to produce attenuation of environmental random noise such as the road traffic noise discussed in the previous chapters. The acoustical design and evaluation problems of barriers have already been considered by many researchers. Almost all of these studies, however, were confined only to the effects on deterministic signals or the gross average evaluation of shielding effects. In this chapter, a method for prediction of the stochastic insulation effect of a sound insulation barrier in the case of arbitrary probability distribution of random noise incident on the barrier is proposed. The emphasis in this chapter is on how the output noise distribution form may be predicted using information on the statistical properties of random input noise and the frequency characteristics of the sound insulation barrier. The effectiveness of

-4-

the proposed prediction method is experimentally confirmed using actual noise data.

In Part II, the relationships between various statistical evaluation indices in relation to  $L_x$  and  $L_{eq}$  for precise evaluation of real environmental random noise are considered. As previously mentioned, two of these indices,  $L_x$  and  $L_{eq}$ , are very important in the field of noise evaluation and regulation. The establishment of a general theory of mutual relationship for these indices is of primary importance for systematic evaluation of environmental random noise. Studies for establishing mutual relationships of this kind have been carried out by many researchers. All most all these studies, however, have been limited to approximate methods derived from the assumption of an ordinary Gaussian distribution and/or practical methods derived by applying the conventional linear regression analysis method to actually observed data. Part II establishes a general theory for mutual relationship between various noise evaluation indices with emphasis on  $L_x$  and  $L_{eq}$ , taking the previously mentioned practical points into consideration.

Chapter 1 deals with a method for estimation of  $L_{eq}$  using the moment statistics of noise level fluctuation of an arbitrary non-Gaussian distribution type. To estimate the value of  $L_{eq}$  for an arbitrary non-Gaussian type random noise fluctuation, a new approach for practical application equivalent to that used for an ordinary Gaussian type is here proposed for the first time. The proposed estimation method is given in a generalized form universally applicable to any kind of random phenomena of a non-Gaussian property, including a

-5-

well-known simplified expression derived under assumption of ordinary Gaussian distribution. Finally, in order to confirm the effectiveness of this method, it is applied to actual road traffic noise data.

Chapter 2 describes the mutual relationships between several statistical indices connected with  $L_x$  and  $L_{eq}$ . A method for estimation of  $L_{eq}$  is presented using the known values of several specific  $L_x$ with the use of the mean and variance of the random noise fluctuation. The proposed estimation formula is also given in a generalized form and includes the well-known simplified expression mentioned above. Next, a new trial for estimation of the original distribution form of the random noise fluctuation is presented using the known values of  $L_{eq}$  and several specific  $L_x$ . The effectiveness of the proposed method is experimentally confirmed by applying it to actual road traffic noise data.

Chapter 3 also describes a method for estimation of the original distribution form of the random noise fluctuation, as discussed in Chapter 2. In this case, a more suitable expression of the probability distribution form is newly introduced in a form matched to the acoustic measurement on a decibel scale. The effectiveness of the proposed method is experimentally confirmed by applying it to actual road traffic noise data, and through comparisons of estimation accuracy between the proposed method and the method described in Chapter 2.

Chapter 4 discusses a precise method for calculation of  $L_x$  and  $L_{eq}$  noise evaluation indices using information on noise level fluctuation (moment statistics or cumulant statistics), based on the previous

-6-

discussion in Chapter 2. Furthermore, a specific on-line measurement system based on this method is constructed with the aid of a microcomputer by introducing an iterative process for extracting moment statistics of random noise fluctuation on a decibel scale. The effectiveness of the proposed method is experimentally confirmed by applying it to actual measurement of road traffic noise.

The conclusion summarizes the results obtained in this study, and future research works on statistical prediction and evaluation of this kind are described. References are given and a list of the author's publications is provided. PART I

1 1

STATISTICAL METHOD FOR PREDICTION OF RANDOM NOISE FLUCTUATION

Chapter 1 A Method for Prediction of Road Traffic Noise Generated from Non-Poisson Type Traffic Flow

#### 1.1 Introduction

Various methods for prediction of road traffic noise have been proposed. (1)-(9) From the methodological point of view, they can be divided into two basic groups, with one group comprising structural methods based on physical models and the other functional methods based on mathematical models. The former is designed to account for various types of physical mechanisms such as traffic flow, propagation characteristic, the geometrical structure of the road, etc. The more such physical mechanisms are taken into account, and the more detailed they are, the more complex the prediction method itself becomes. The latter, however, is constructed by pre-establishing a mathematical model for the actual noise fluctuation pattern itself. If only the latter method is used in the prediction of road traffic noise, it is not possible to infer quantitatively physical clues to practical noise control improvement. Accordingly, it would be effective to introduce some kind of hybrid method combining the latent advantages of both methods of prediction.(8)

In actual traffic flow, a general, arbitrary non-Poisson type flow distribution is more often seen than the idealized Poisson type, since traffic conditions are rarely ideal. Furthermore, actual road

-8-

traffic noises exhibit various types of probability distributions, due to the diversified causes of the noise fluctuations.

In this chapter, a practical method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow is proposed. The method is based on an approach equivalent to that used for an idealized Poisson type traffic flow. It allows for treatment of any kind of road traffic noise fluctuation, provided that information on the actual response wave form generated by the surrounding sound propagation environment of a level time pattern observed when one vehicle passes directly in front of an observation point can first be obtained.

The effectiveness of the proposed prediction method has been experimentally confirmed by applying it to observed road traffic noise data.

#### 1.2 Theoretical consideration

1.2.1 Description of problem

Considering the wave form of noise level fluctuation at an observation point, as shown in Fig.1.1.1, let  $w(\xi_i)(\xi_i \Delta t - \tau_i)$  be the normalized response wave form w(0)=1 observed when one vehicle passes directly in front of an observation point during a specific time



Fig.1.1.1 Time pattern of the noise level fluctuation generated by vehicles passing directly in front of an observation point. interval, T. It is supposed that the ith vehicle generates the noise intensity  $Y_i$  at a time  $t=\tau_i$ . At this time, - based on the additive property of energy quantities, the total noise intensity, x(t), at a time t is directly expressed as

x (t) = 
$$\sum_{i=1}^{N} Y_i w (t - \tau_i)$$
, (1.1.1)

where N denotes the number of vehicles passing directly in front of an observation point during the time interval T. It is quite natural to assume that the random variable  $Y_i$  is mutually independent of the random variable  $\tau_i$ .

The problem is how to establish a practical hybrid method for predicting the noise level probability distribution by estimating the statistics of this noise intensity fluctuation, as related to the actual traffic flow and the elementary response wave form  $w(\xi_i)$ .

# 1.2.2 Arbitrary non-linear function type statistics for an arbitrary non-Poisson type traffic flow

The main purpose of this section is to present an explicit derivation of an expression for an arbitrary non-linear function type expectation value which is equivalent to that for a well-known Poisson type probability function. This will provide the advantage of being able to use the technical properties of Poisson's law. Let  $z_i (\in [z_L, z_M])$  be the ith quantized amplitude with an amplitude difference in-

terval h. According to the previous work,<sup>(10)</sup> the probability function  $P(z_i)$  for a quantized random variable  $z_i$  can be generally expressed in a statistical series expansion form, as

$$P(z_i) = \sum_{n=0}^{\infty} b_n \left(\frac{d}{h}\right)^n \nabla^n P_x(z_i) \qquad (1.1.2)$$

with

$$b_{n} \triangleq \frac{(-1)^{n}}{n!} \langle (z_{i} - x_{i})^{(n)} \rangle_{zi} , \qquad (1.1.3)$$

where  $P_{x}(x_{i})$  is the basic probability function for a quantized random variable  $x_{i}$  with an amplitude difference interval h, which can be, artificially or arbitrarily, established in advance from the operational viewpoint. In Eq.(1.1.2),  $\langle \cdot \rangle_{z_{i}}$  denotes an expectation operation with respect to  $z_{i}$  and  $(.)^{(n)}$  denotes the factorial function defined by

$$\xi_{i}^{(n)} \triangleq \xi_{i} (\xi_{i} - h) \cdot \cdot \cdot [\xi_{i} - (n - 1) h] \quad (n \ge 1),$$

$$\xi_{i}^{(0)} \triangleq 1$$
. (1.1.4)

 $\triangledown$  denotes the backward difference operator :

$$\nabla f(z) \triangleq (1/d) [f(z) - f(z - h)]$$
 (1.1.5)

with two arbitrary constants, d and h. In Eq.(1.1.5), the constant d is introduced for the purpose of relating the discrete amplitude type expression (d=1), and the continuous amplitude type expression (d=h  $\rightarrow$  0). Thus, after introducing an arbitrary non-linear function g(.) with respect to  $z_i$ , its expectation value is directly given by using Eq.(1.1.2), as

$$\sum_{i=L}^{M} g (z_i) P (z_i) = \sum_{n=0}^{\infty} b_n \left(\frac{d}{h}\right)^n \sum_{i=L}^{M} g (z_i) \nabla^n P_x (z_i)$$
$$= \sum_{n=0}^{\infty} b_n \left(\frac{d}{h}\right)^n (-1)^n \sum_{i=L}^{M} \Delta^n g (z_i) P_x (z_i) . \quad (1.1.6)$$

Hereupon, the relationship :

$$G (z_i) \nabla F (z_i) = \triangle \left\{ G (z_i) F (z_{i-1}) \right\}$$

$$- F (z_i) \triangle G (z_i)$$
(1.1.7)

and a new type forward difference operation :

$$\Delta f(z) \triangleq (1/d) [f(z+h) - f(z)]$$
 (1.1.8)

( $\Delta$  is the forward difference operator) have been used.

Upon employing, respectively, a quantized random variable  $z_i$ , the basic probability function  $P_x(.)$ , with the number of vehicles being denoted by  $N(\in[0,M])$  and a Poisson type probability function  $^{(11),(12)}$ :

$$P_{\mathbf{x}}(N) = \frac{1}{N!} e^{-\lambda} \lambda^{N}$$
 (1.1.9)

$$A \triangleq \langle N \rangle_{N}, \qquad (1.1.10)$$

Eq.(1.1.6) can easily be rewritten as

$$\langle g (N) \rangle_{N} \left( \triangleq \sum_{N=0}^{M} g (N) P (N) \right)$$

$$= \sum_{n=0}^{\infty} b_{n} (-1)^{n} \sum_{N=0}^{M} \triangle^{n} g (N) P_{x} (N)$$

$$= \left\langle \sum_{n=0}^{\infty} b_{n} (-1)^{n} \triangle^{n} g (N) \right\rangle_{x}, \quad (1.1.11)$$

for traffic flow with an arbitrary type probability function P(N). Here, the two arbitrary constants are taken to be h=d=1 and <.><sub>x</sub> denotes an expectation based on a Poisson type probability function  $P_x(N)$ . The expansion coefficient  $b_n$  is given by <sup>(10)</sup>

$$b_{n} = \frac{(-1)^{n}}{n!} \left\{ \langle N^{(n)} \rangle_{N} - \sum_{k=0}^{n-1} (-1)^{k} \frac{n!}{(n-k)!} b_{k} \lambda^{n-k} \right\}$$
  
$$b_{0} = 1. \qquad (1.1.12)$$

From Eq.(1.1.11), it should be noticed that the expectation value  $\langle g(N) \rangle_N$  in terms of an arbitrary non-Poisson type probability function can be equivalently calculated based on an expectation operation in terms of an ordinary Poisson type probability function.

with

1.2.3 Prediction of road traffic noise by use of moment statistics for the noise intensity fluctuation

The conditional moment generating function  $\langle e^{\theta X} | N \rangle_X$  with respect to the noise intensity fluctuation x(t) can be regarded as an arbitrary non-linear function form  $g(N, \theta)$ . That is, all the characteristics of the non-Poisson distributions can be related to the conditional moment generating function  $\langle e^{\theta X} | N \rangle_X$  thus producing the moment statistics  $\langle x^m \rangle$ :

$$g(N, \theta) \triangleq \langle e^{\theta \mathbf{x}} | N \rangle_{\mathbf{x}}.$$
 (1.1.13)

The moment generating function  $m(\theta)(\underline{\Delta}^{<}\exp(\theta x)^{>})$  can be directly derived by use of a relation equivalent to an ordinary Poisson type in Eq.(1.1.11), as follows :

$$\mathbf{m} (\theta) = \langle \langle e^{\theta \mathbf{x}} | \mathbf{N} \rangle_{\mathbf{x}} \rangle_{\mathbf{N}}$$

$$= \sum_{\mathbf{n}=\mathbf{0}}^{\infty} \mathbf{b}_{\mathbf{n}} (-1)^{\mathbf{n}} \sum_{\mathbf{N}=\mathbf{0}}^{\infty} \Delta^{\mathbf{n}} \langle e^{\theta \mathbf{x}} | \mathbf{N} \rangle_{\mathbf{x}} \frac{1}{-\mathbf{N}} e^{-\lambda} \lambda^{\mathbf{N}}.$$

$$(1.1.14)$$

By substituting the difference calculation based on Eq.(1.1.1) :

$$= \langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}}^{\mathbf{N}} (\langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}} - 1)^{\mathbf{n}}$$
(1.1.15)

into Eq.(1.1.14), one can subsequently obtain

$$m (\theta) = \sum_{n=0}^{\infty} b_n (-1)^n (\langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}} - 1)^n$$
$$\cdot e^{-\lambda} \sum_{\mathbf{N}=0}^{\infty} \frac{1}{\mathbf{N}!} \langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}}^{\mathbf{N}} \lambda^{\mathbf{N}}.$$
(1.1.16)

Furthermore, by using two calculation procedures based on a Taylor series expansion :

$$(\langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}} - 1)^{\mathbf{n}} = \left( \sum_{m=1}^{\infty} \frac{\theta^{m}}{m!} \langle \mathbf{Y}^{m} \rangle \langle \mathbf{w}^{m} \rangle \right)^{\mathbf{n}},$$

$$e^{-\lambda} \sum_{\mathbf{N}=0}^{\infty} \frac{1}{\mathbf{N}!} \langle e^{\theta \mathbf{Y} \mathbf{w}} \rangle_{\mathbf{Y}, \mathbf{w}}^{\mathbf{N}} \lambda^{\mathbf{N}}$$

$$= e \times p \left\{ \lambda \sum_{m=1}^{\infty} \frac{\theta^{m}}{m!} \langle \mathbf{Y}^{m} \rangle \langle \mathbf{w}^{m} \rangle \right\},$$

$$(1.1.17)$$

the moment generating function  $m(\,\theta\,)$  can be consequently obtained as follows :

$$m (\theta) = m_{0} (\theta) \sum_{n=0}^{\infty} b_{n} (-1)^{n} \left( \sum_{m=1}^{\infty} \frac{\theta^{m}}{m!} \langle Y^{m} \rangle \langle W^{m} \rangle \right)^{n}$$
$$= \sum_{n=0}^{\infty} b_{n} (-1)^{n} \frac{\partial^{n}}{\partial \lambda^{n}} m_{0} (\theta) \qquad (1.1.18)$$

with

$$\mathbf{m}_{\mathbf{0}} (\theta) \triangleq \mathbf{e} \times \mathbf{p} \left\{ \lambda \sum_{\mathbf{m}=1}^{\infty} \frac{\theta^{\mathbf{m}}}{\mathbf{m} !} \langle \mathbf{Y}^{\mathbf{m}} \rangle \langle \mathbf{w}^{\mathbf{m}} \rangle \right\} .$$
(1.1.19)

Needless to say,  $m_0(\theta)$  denotes the moment generating function in the case of an idealized Poisson type traffic flow. Thus, the objective

moment generating function  $m(\theta)$  can be given by simply differentiating  $m_0(\theta)$  by the parameter  $\lambda$ . From the above equation, the mth order moment of x(t) can easily be obtained as

$$\langle \mathbf{x}^{\mathbf{m}} \rangle \triangleq \frac{\partial^{\mathbf{m}}}{\partial \theta^{\mathbf{m}}} \mathbf{m} (\theta) \bigg|_{\theta=0} = \sum_{\mathbf{n}=0}^{\infty} \mathbf{b}_{\mathbf{n}} (-1)^{\mathbf{n}} \frac{\partial^{\mathbf{n}}}{\partial \lambda^{\mathbf{n}}} \langle \mathbf{x}^{\mathbf{m}} \rangle_{\mathbf{0}},$$
<sup>(1.1.20)</sup>

where  $\langle x^m \rangle_0$  is the mth order moment of x for the case of an idealized Poisson type traffic flow. Furthermore, since the mth order cumulant of x for the case of an idealized Poisson type traffic flow is given by

$$\kappa_{\mathbf{m}\mathbf{0}} = \lambda \quad (\mathbf{Y}^{\mathbf{m}}) \quad (\mathbf{w}^{\mathbf{m}}(\boldsymbol{\xi})) \qquad (1.1.21)$$

with

$$\langle w^{\mathbf{m}}(\xi) \rangle = \frac{1}{T} \int_{-T/2}^{T/2} w^{\mathbf{m}}(\xi) d\xi$$

(:: $\tau_i$ : uniform distribution)

(1.1.22)

the following two conditions are derived :

$$(\partial / \partial \lambda) \kappa_{m0} = \langle Y^{m} \rangle \langle W^{m} (\xi) \rangle,$$
  
 $(\partial^{n} / \partial \lambda^{n}) \kappa_{m0} = 0 \qquad (n \ge 2).$  (1.1.23)

By considering these two conditions and using the mathematical rela-

tionship between the moment statistics and the cumulant statistics, Eq.(1.1.20) can be expressed explicitly as

$$(x) = (x) \circ - b_{1} \frac{\partial}{\partial \lambda} \kappa_{10},$$

$$(x^{2}) = (x^{2}) \circ - b_{1} \left\{ \frac{\partial}{\partial \lambda} \kappa_{20} + 2 \langle x \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{10} \right\} + 2 b_{2} \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{2},$$

$$(x^{3}) = (x^{3}) \circ - b_{1} \left\{ \frac{\partial}{\partial \lambda} \kappa_{30} + 3 \langle x \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{20} + 3 \langle x^{2} \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{10} \right\}$$

$$+ 6 b_{2} \left\{ \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right) \left( \frac{\partial}{\partial \lambda} \kappa_{20} \right) + \langle x \rangle \circ \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{2} \right\}$$

$$- 6 b_{3} \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{3},$$

$$(x^{4}) = (x^{4}) \circ - b_{1} \left\{ 4 \langle x^{3} \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{10} + 6 \langle x^{2} \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{20}$$

$$+ 4 \langle x \rangle \circ \frac{\partial}{\partial \lambda} \kappa_{30} + \frac{\partial}{\partial \lambda} \kappa_{40} \right\}$$

$$+ b_{2} \left\{ 1 2 \langle x^{2} \rangle \circ \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{2} + 2 4 \langle x \rangle \circ \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right) \left( \frac{\partial}{\partial \lambda} \kappa_{20} \right)$$

$$+ 6 \left( \frac{\partial}{\partial \lambda} \kappa_{20} \right)^{2} + 8 \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right) \left( \frac{\partial}{\partial \lambda} \kappa_{30} \right) \right\}$$

-18-

0

$$-1 \ 2 \ b \ 3 \left\{ 3 \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{2} \left( \frac{\partial}{\partial \lambda} \kappa_{20} \right) - 4 \ \langle x \rangle \ 0 \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{3} \right\}$$
$$+ 2 \ 4 \ b \ 4 \left\{ \left( \frac{\partial}{\partial \lambda} \kappa_{10} \right)^{4} \right\} , \dots . \qquad (1.1.24)$$

In these equations, the expansion coefficient  $b_n$  (n=1,2,...) is explicitly expressed from its definition, as

$$b_{0} = 1, \qquad b_{1} = -\langle N \rangle_{N} + \lambda,$$

$$b_{2} = \langle 1/2 \rangle \langle N \rangle_{N} + 2 \langle N \rangle_{N} + \lambda^{2},$$

$$b_{3} = \langle 1/3 \rangle \langle -\langle N \rangle_{N} + 3 \langle N \rangle_{N} + \lambda^{2}, \lambda^{2} + \lambda^{3},$$

$$b_{4} = \langle 1/4 \rangle \langle N \rangle_{N} + 2 \langle N \rangle_{$$

By using the additive property of cumulant statistics with respect to the independent random variables, one can extend the above theory to cases in which several types of vehicle pass on a multi-laned road.

Therefore, by using the above estimated values for moment statistics  $\langle x^m \rangle$  (m=1,2,...), the cumulative distribution function Q(x) with respect to x(t) can be expressed in the form of a statistical Laguerre series expansion universally applicable to arbitrary distribution types, as<sup>(13)</sup>

Q (x) = 
$$\int_{0}^{x/s} P_{r}(u;m) du$$

-19-

$$+ P_{\Gamma}\left(\frac{x}{s}; m+1\right) \sum_{n=3}^{\infty} A_{n}L_{n-1}(m)\left(\frac{x}{s}\right) \qquad (1.1.26)$$

with

$$m = \frac{\langle x \rangle^{2}}{\langle (x - \langle x \rangle)^{2} \rangle}, \qquad s = \frac{\langle (x - \langle x \rangle)^{2} \rangle}{\langle x \rangle}$$

$$A_{n} \triangleq \frac{(n-1)!\Gamma(m+1)}{\Gamma(m+n)} \left\langle L_{n}^{(m-1)} \left( \frac{x}{s} \right) \right\rangle$$
$$= \sum_{k=0}^{n} \frac{(-1)^{k}}{k!} \frac{(n-1)!\Gamma(m+1)}{\Gamma(n-k+1)\Gamma(m+k)} \frac{\langle x^{k} \rangle}{s^{k}}, \quad (1.1.27)$$

where  $P_{r}(u;m)$  is the well-known Gamma distribution defined by

$$P_{\Gamma}(u;m) \triangleq [1/\Gamma(m)] e^{-u} u^{m-1}.$$
 (1.1.28)

Each order of the moment statistics  $\langle x^m \rangle$  (m=1,2,...,n) is thus first obtained by Eq.(1.1.24) with the use of the expansion coefficients b<sub>m</sub> (m=1,2,...,n) which characterize the non-Poisson properties. The estimated values based on each of these moment statistics  $\langle x^m \rangle$  (m=1,2,..., n) thus influence each expansion coefficient A<sub>m</sub> (m=1,2,...,n), as shown in Eq.(1.1.27). Therefore, each expansion coefficient A<sub>m</sub> in the statistical Laguerre series expansion, Eq.(1.1.26) corresponds to the non-Poisson distribution expansion coefficient,  $b_i$   $(1 \le i \le m)$  in Eq.(1.1.25).

To sum up, the moment statistics  $\langle x^m \rangle$  closely related to the non-Poisson properties influence each expansion coefficient  $A_m$  in the cumulative distribution function Q(x), expressed in the form of a statistical Laguerre series expansion, Eq.(1.1.26).

1.3 Experimental consideration

1.3.1 Simplification of an elementary response wave form by use of triangular shape approximation<sup>(14)</sup>

The experimental procedures can be simplified by using a triangular shape to approximate an elementary response wave form due to the surrounding sound propagation environment when one vehicle passes directly in front of an observation point during the time interval T. For the purpose of determining  $\langle x^m \rangle$ , it is not necessary to accurately determine w( $\xi$ ) itself. Only the resultant value of the definite integral  $\int_{-T/2}^{T/2} w^m(\xi) d\xi$  need be obtained. The values of the cumulant statistics of the noise intensity fluctuation are not particularly sensitive to the exact wave form of w( $\xi$ ), due to the effects of various types of smoothing operations.

The approximation method based on a triangular shape is as follows :

(1) The elementary response wave form of the actual time pattern is

first approximated by a triangular shape with the inclination  $\alpha$  (dB/second) so that the total intensity of the approximated wave form is equal to that of the actual one, as shown in Fig.1.1.2.

(2) After this approximation procedure, the approximated wave form on a decibel scale is transformed into a normalized wave form on an intensity scale, as shown in Fig.1.1.3, the approximated wave form  $w(\xi)$  of the actual response wave thus given beeing

$$w (\xi) = e^{-\alpha |\xi|/M} \qquad (M \triangleq 1 \ 0 \ / 1 \ n \ 1 \ 0) \ . \tag{1.1.29}$$

#### 1.3.2 Predicted results based on the proposed method

The proposed prediction method has been applied to road traffic noise data observed near a national main road with two lanes (an up-lane and a down-lane). The arrangements for measuring road traffic noise at an observation point is shown in Fig.1.1.4. In this experiment, the passing vehicles are classified into two categories : heavy and light. The numbers of vehicles in each lane passing directly in front of an observation point within time intervals of 5 minutes were counted. The parameter  $\lambda$  was directly obtained by averaging the number of vehicles. The values of the moment statistics  $\langle Y^{m} \rangle$  were obtained by averaging peak values of observed elementary response wave forms of noise intensities generated from light and heavy vehicles.



Fig.1.1.2 Simplified treatment of the actual wave form on a decibel scale by use of the triangular approximation.



Fig.1.1.3 Transformation of an approximated response wave form on a decibel scale into a normalized response wave form on an intensity scale.



Fig.1.1.4 Arrangements for measuring road traffic noise at an observation point.

Figure 1.1.5, from up-lane traffic flow data, shows a comparison between the theoretically estimated curves and the experimentally sampled points for the cumulative distribution form of the number of passing vehicles. In this figure, the two theoretical curves presented are obtained using only the well-known Poisson type probability function as the first expansion term, and the arbitrary non-Poisson type probability function in Eq. (1.1.2). These theoretically estimated curves obtained by use of Eq.(1.1.9) and the second approximation of Eq.(1.1.2) are shown by the full and dashed lines, respectively. The estimated results for the down-lane are shown in Fig.1.1.6. From these results, it can be seen that the curves estimated with only an idealized Poisson type probability function do not agree well with the experimentally sampled points : that is, the traffic flows in both lanes clearly display statistical properties of non-Poisson type.

The observed noise data and two sets of traffic flow data (upand down lanes) were used to obtain the results shown in Fig.1.1.7. This is a comparison between the theoretical curves predicted under the approximate assumption of an idealized Poisson type traffic flow and that obtained from the experimentally sampled points, for the cumulative distribution form of the A-weighted noise level fluctuation. A comparison between the theoretically predicted curves obtained by use of the proposed method based on the non-Poisson type traffic flow and that obtained from experimentally sampled points for the cumulative distribution form of the A-weighted noise level fluctu-

-25-



Fig.1.1.5 A comparison between experimentally sampled points and theoretically estimated curves obtained by using the unified series expansion expression(see Eq.(1.1.2)) for the cumulative distribution form of the number of passing vehicles on the up-lane of the road. Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ———, the first approximation corresponding to an idealized Poisson type probability function; ----, the third approximation.






Fig.1.1.7 A comparison between experimentally sampled points and theoretically predicted curves obtained by using only an idealized Poisson type traffic flow for the cumulative distribution form of the noise level fluctuation. Experimentally sampled points are marked by  $\bullet$  and theoretically predicted curves are respectively shown as \_\_\_\_\_, the first approximation of Eq.(1.1.26); ----, the second approximation (and the third approximation).

ation is shown in Fig.1.1.8. It is obvious that the successive addition of higher order expansion terms moves the theoretically predicted curves closer to the experimentally sampled points. However, as shown in Fig.1.1.7, the theoretically predicted curves based only on an idealized Poisson type traffic flow do not agree well with the experimentally sampled points, even when the number of correction terms is increased. These figures show clearly that the prediction accuracy of the proposed method is much better than that of simplified prediction methods in which only idealized Poisson type traffic flow is used. It should be noted that the prediction errors of the evaluation indices  $L_5$ ,  $L_{10}$  and  $L_{50}$  usually used in noise evaluation and regulation problems are mostly within ±1 dB for the proposed prediction method.

#### 1.4 Conclusion

This study has focussed on the establishment of a hybrid method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow, with emphasis on the methodological point of view.

Firstly, expressions for the statistical moments, of an arbitrary non-linear function type, for this arbitrary non-Poisson type traffic flow have been presented, as obtained by an approach equivalent to that used for an idealized Poisson type traffic flow. A unified explicit expression of series expansion type was then introduced



Fig.1.1.8 A comparison between experimentally sampled points and the theoretically predicted curves obtained by using the proposed method based on arbitrary non-Poisson type traffic flow for the cumulative distribution form of the noise level fluctuation. Experimentally sampled points are marked by • and theoretically predicted curves are respectively shown as \_\_\_\_\_, the first approximation of Eq. (1.1.26); ----, the second approximation; \_--, the third approximation.

for vehicle number distribution in an arbitrary non-Poisson type traffic flow. Next, a unified statistical prediction method for road traffic noise, based on introduction of an elementary response wave form, as influenced by the surrounding sound propagation environment when one vehicle passes directly in front of an observation point, was proposed in a general explicit expression form of series expansion type. Finally, the effectiveness of the proposed prediction method has been experimentally confirmed by applying it to observed road traffic noise data.

This research is still in the early stages and the work reported here has focussed on principally on methodological aspects. Problems remain in the application of the proposed prediction method to other actual engineering situations. Chapter 2 A Method for Prediction of Road Traffic Noise at a T-Type Road Intersection Based on the Image Method

2.1 Introduction

In Chapter 1, a general method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow was proposed, based on information on the elementary response wave form when one vehicle passes directly in front of an observation point. In order to establish a more effective prediction method, a theoretical method for estimating the above-mentioned elementary response wave form in complicated realistic acoustical environments must be found. In this chapter, a new approach for estimating theoretically this elementary response wave form at a T-type road intersection with sound insulation barriers is proposed, using a well-known image method.<sup>(15),(16)</sup> The effectiveness of the proposed method has been experimentally confirmed by applying it to the road traffic noise data observed at a Ttype road intersection in a city area.

2.2 Theoretical consideration

2.2.1 Outline of prediction theory

Let us consider the wave form of the noise intensity fluctuation x(t) at an observation point, as stated in Chapter 1. By substituting

Eq.(1.1.25) into Eq.(1.1.24) in Chapter 1, the moment statistics of x(t) can be resultantly expressed as  $\langle x \rangle = \langle N \rangle \langle Y \rangle \langle w (\xi) \rangle ,$   $\langle x^2 \rangle = \langle N (N-1) \rangle \langle Y \rangle^2 \langle w (\xi) \rangle^2 + \langle N \rangle \langle Y^2 \rangle \langle w^2 (\xi) \rangle ,$   $\langle x^3 \rangle = \langle N (N-1) (N-2) \rangle \langle Y^3 \rangle \langle w (\xi) \rangle^3 + 3 \langle N (N-1) \rangle \langle Y^2 \rangle \langle Y \rangle \langle w^2 (\xi) \rangle \langle w (\xi) \rangle + \langle N \rangle \langle Y^3 \rangle$   $\cdot \langle w^3 (\xi) \rangle ,$   $\langle x^4 \rangle = \langle N (N-1) (N-2) (N-3) \rangle \langle Y^4 \rangle \langle w (\xi) \rangle ^4$   $+ 6 \langle N (N-1) (N-2) \rangle \langle Y^2 \rangle \langle Y \rangle^2 \langle w^2 (\xi) \rangle$   $\cdot \langle w (\xi) \rangle^2 + 4 \langle N (N-1) \rangle \langle Y^3 \rangle \langle Y \rangle \langle w^3 (\xi) \rangle$   $\cdot \langle w (\xi) \rangle + 3 \langle N (N-1) \rangle \langle Y^2 \rangle^2 \langle w^2 (\xi) \rangle^2 + \langle N \rangle$  $\cdot \langle Y^4 \rangle \langle w^4 (\xi) \rangle , \cdots$  (1.2.1)

Based on the additive property of cumulant statistics with respect to the independent random variables, one can extend this theory to cases in which several types of vehicle pass on a multi-laned road, as mentioned in Chapter 1. The above moment statistics  $\langle x^m \rangle$  on an intensity scale can be transformed into the cumulant statistics  $\kappa_{Lm}$ on a decibel scale by using the following relationship<sup>(8)</sup>:

$$\frac{1}{\mathbf{x}_{0}^{\mathbf{m}}} \langle \mathbf{x}^{\mathbf{m}} \rangle = \mathbf{e} \mathbf{x} \mathbf{p} \left\{ \sum_{i=1}^{\infty} \frac{1}{i!} \left( \frac{\mathbf{m}}{\mathbf{M}} \right) \kappa_{\mathbf{L}i} \right\}.$$

$$(\mathbf{x}_{0} \triangleq 1 \ 0^{-12} \mathbf{W} / \mathbf{m}^{2}, \qquad \mathbf{M} \triangleq 1 \ 0 \ / \ 1 \ \mathbf{n} \ 1 \ 0 ) \qquad (1.2.2)$$

Thus, after regarding the cumulant statistics  $\kappa_{Li}$  as unknown parameters and by solving the simultaneous equations derived from Eq.(1.2. 2) (i,m=1,2,...,M), the cumulant statistics  $\kappa_{Li}$  (i=1,2,...,M) on a decibel scale can be obtained. For example, the transformation formula for the cumulant statistics up to the fourth order is explicitly expressed by

$$\kappa_{L1} = M \left\{ 4 \ 1 \ n \ \frac{\langle x \rangle}{x_{0}} - 3 \ 1 \ n \ \frac{\langle x^{2} \rangle}{x_{0}^{2}} + \frac{4}{3} \ 1 \ n \ \frac{\langle x^{3} \rangle}{x_{0}^{3}} - \frac{1}{4} \ 1 \ n \ \frac{\langle x^{4} \rangle}{x_{0}^{4}} \right\} ,$$

$$\kappa_{L2} = M^{2} \left\{ -\frac{2 \ 6}{3} \ 1 \ n \ \frac{\langle x \rangle}{x_{0}^{4}} + \frac{1 \ 9}{2} \ 1 \ n \ \frac{\langle x^{2} \rangle}{x_{0}^{2}} - \frac{1 \ 4}{3} \ 1 \ n \ \frac{\langle x^{3} \rangle}{x_{0}^{3}} + \frac{1 \ 1}{1 \ 2} \ 1 \ n \ \frac{\langle x^{4} \rangle}{x_{0}^{4}} \right\} ,$$

$$\kappa_{L3} = M^{3} \left\{ 9 \ 1 \ n \ \frac{\langle x \rangle}{x_{0}} - 1 \ 2 \ 1 \ n \ \frac{\langle x^{2} \rangle}{x_{0}^{2}} + 7 \ 1 \ n \ \frac{\langle x^{3} \rangle}{x_{0}^{3}} - \frac{3}{2} \ 1 \ n \ \frac{\langle x^{4} \rangle}{x_{0}^{4}} \right\} .$$

$$\kappa_{L4} = M^{4} \left\{ -4 \ 1 \ n \ \frac{\langle x \rangle}{x_{0}} + 6 \ 1 \ n \ \frac{\langle x^{2} \rangle}{x_{0}^{2}} - 4 \ 1 \ n \ \frac{\langle x^{3} \rangle}{x_{0}^{3}} + 1 \ n \ \frac{\langle x^{4} \rangle}{x_{0}^{4}} \right\} .$$
(1.2.3)

In order to evaluate the  $L_x$  noise evaluation indices (e.g.,  $L_5, L_{10}$ ,  $L_{50}$  and  $L_{90}$ ), the cumulative distribution function Q(L) of the noise level fluctuation L can be expressed in an orthonormal series expansion, as a statistical Hermite series expansion universally applicable to an arbitrary non-Gaussian distribution form, as follows <sup>(14)</sup>:

$$Q (L) = \frac{1}{\sqrt{2 \pi}} \int_{-\infty}^{\frac{L-\kappa}{\kappa} + \frac{1}{2}} e^{\frac{z}{2}} d\xi$$
$$- \sum_{n=3}^{\infty} B_n (\kappa_L) \frac{1}{\sqrt{2 \pi}} e^{x} p \left\{ (L - \kappa_{L1})^2 / 2 \kappa_{L2} \right\}$$
$$\cdot H_{n-1} \left( \frac{L - \kappa_{L1}}{\sqrt{\kappa_{L2}}} \right) \qquad (1.2.4)$$

with

$$B_{n}(\kappa_{L}) \triangleq \frac{1}{n!} \left\langle H_{n} \left( \frac{L - \kappa_{L1}}{\sqrt{\kappa_{L2}}} \right) \right\rangle , \qquad (1.2.5)$$

where  $H_n(.)$  denotes the nth order Hermite polynomial.

2.2.2 Estimation of the elementary response wave form using the image method

Let us consider the sound propagation characteristic of a T-type road intersection as shown in Fig.1.2.1. There are homogeneous sound insulation barriers made of concrete material along both sides of the road. It is assumed that each vehicle drives in the middle of each lane with a proper constant speed, and the area across the river is regarded as a free field of sound. Figure 1.2.2 shows the geometrical construction of the T-type road intersection under consideration. First, an explicit expression for the sound propagation characteristic at each vehicle position can be derived by using a (15), (16)The elementary response wave form well-known image method. in the time domain can then be evaluated by taking this sound propagation characteristic into consideration, together with the average speed of the passing vehicles. In this case, it was possible to assume that diffraction effects from high sound insulation barriers is negligibly small, through the well-known evaluation of sound insulation barriers using Maekawa's chart with a Fresnel number. Let



Fig.1.2.1 Arrangements for measuring road traffic noise at an observation point.



1

Fig.1.2.2 Geometrical construction of the T-type road intersection under consideration.

w(z) be the sound propagation characteristic at an arbitrary vehicle position z. Here, w(z) is classified into the following three categories (corresponding to each vehicle position  $S_d$ ,  $S_h$  and  $S_k$  in Fig.1.2.2).

[A] Sound propagation characteristic  $w_d(z)$  due to the direct sound and the sound reflected from the road surface

Let  $w_d(z)$  be the sound propagation characteristic due to the direct sound and the sound reflected from the road surface, related to a vehicle position  $S_d$  in Fig.1.2.2. Figure 1.2.3 shows its schematic analysis for the sound propagation characteristic  $w_d(z)$ . Based on the geometrical relationship among the sound source  $S_d$  (vehicle position), the receiver 0 (observation point) and the image receiver 0', with the aid of the additive property of energy quantities,  $w_d(z)$  is given by

$$w_{d}(z) = D(z_{d2}-z) D(z-z_{d1})$$

$$\cdot \frac{1}{4\pi} \left\{ \frac{1}{X^{2}+(z-a)^{2}+h^{2}} + \frac{R_{1}}{X^{2}+(z-a)^{2}+(2h_{1}-h)^{2}} \right\},$$
(1.2.6)

where D(.) is the truncation function defined as

$$D(\zeta) \triangleq \begin{cases} 1 & (\zeta > 0), \\ 0 & (\zeta \le 0). \end{cases}$$
(1.2.7)

In Eq.(1.2.6), h denotes the difference in height between the sound source and the receiver O, and  $R_1$  denotes the reflection coefficient



a) Schematic analysis of the direct sound.



b) Schematic analysis of sound reflection from the road surface.

Fig.1.2.3 Schematic analysis for the sound propagation characteristic  $w_{d}(z)\,.$ 

of the road surface. In Fig.1.2.3,  $z_{d1}$  and  $z_{d2}$  denote respectively the minimum and maximum values of z at which direct sound and reflected sound from the road surface can be received :

$$z_{d1} = -\frac{L_{z}}{2} \frac{X}{X_{1}} - a\left(\frac{X}{X_{1}} - 1\right) ,$$

$$z_{d2} = \frac{L_{z}}{2} \frac{X}{X_{1}} - a\left(\frac{X}{X_{1}} - 1\right) .$$
(1.2.8)

[B] Sound propagation characteristic  $w_h(z)$  based on the sounds reflected from the barriers on both sides of the road

In this case, one can apply a well-known image method to two specific cases when the first reflection occurs on the barriers either near to or away from the observation point. In the former case, the sound propagation characteristic due to reflections from the barriers is denoted by  $w_{h1}(z,n)$ . That of the latter case is denoted by  $w_{h2}(z,n)$ . Figure 1.2.4 shows a schematic analysis of sound reflections from the barriers. Based on a geometrical analysis of n reflections from the barriers (denoted by  $0^n$ ) through image receivers, one can easily obtain

$$w_{h}(z) = \sum_{n=1}^{\infty} [w_{h1}(z, n) + w_{h2}(z, n)]$$
$$= \sum_{n=1}^{\infty} \left\{ D(z_{h12} - z) D(z + z_{h11}) \right\}$$

-39-



Fig.1.2.4 Schematic analysis of sound reflections from the sound insulation barriers.

$$\cdot \frac{1}{4 \pi} \left[ \frac{R_{2}^{n}}{X^{2} + (n L_{z} + (-1))^{n} a + z)^{2} + h^{2}} \right]$$

$$+ D (z_{h22} - z) D (z + h_{21})$$

$$\cdot \frac{1}{4 \pi} \left[ \frac{R_{2}^{n}}{X^{2} + (n L_{z} + (-1))^{n} a + z)^{2} + h^{2}} \right] \right\} , \quad (1.2.9)$$

where  $R_2$  denotes the reflection coefficient of the barriers. Moreover,  $z_{h11}$ ,  $z_{h12}$ ,  $z_{h21}$  and  $z_{h22}$  are the minimum and maximum values of z at which reflected sounds for  $w_{h1}(z,n)$  and  $w_{h2}(z,n)$  can be received :

$$z_{h11} = -(n L_{z} + (-1)^{n} a) \left(\frac{X}{X_{0}} - 1\right) - \left(\frac{L_{z}}{2} + b\right) \frac{X}{X_{0}},$$

$$z_{h12} = -(n L_{z} + (-1)^{n} a) \left(\frac{X}{X_{0}} - 1\right) + \frac{L_{z}}{2} \frac{X}{X_{1}},$$

$$z_{h21} = (n L_{z} + (-1)^{n} a) \left(\frac{X}{X_{0}} - 1\right) - \frac{L_{z}}{2} \frac{X}{X_{1}},$$

$$z_{h22} = (n L_{z} + (-1)^{n} a) \left(\frac{X}{X_{0}} - 1\right) + \left(\frac{L_{z}}{2} + b\right) \frac{X}{X_{0}}.$$
(1.2.10)

[C] Sound propagation characteristic  $w_{\rm K}(z)$  due to sound reflection from the corner of the road intersection

Let  $w_{k1}(z,n)$  be the sound propagation characteristic for a case in which the second reflection occurs at the barrier near the observation point after the first reflection at the corner, then, n reflections occur on the barriers. Alternatively, let  $w_{k2}(z,n)$  be the sound propagation characteristic for a case in which the second reflection occurs at the barrier away from the same observation point. By consideration of both the image receivers and the image sources as shown in Fig.1.2.5 ( $S'_{k21}$  and  $S'_{k22}$ ), one can obtain :

$$w_{k}(z) = \sum_{n=1}^{\infty} [w_{k1}(z, n) + w_{k2}(z, n)]$$

$$= \sum_{n=1}^{\infty} \left\{ D((z_{k12} - z)) d((z + z_{k11})) \\ \cdot \frac{1}{4\pi} \left[ \frac{R 2^{n+1}}{(X + z - x_{2})^{2} + (X_{2} + n L_{z} + (-1))^{n} a)^{2} + h^{2}} \right] \\ + D((z_{k22} - z)) D((z + z_{k21})) \\ \cdot \frac{1}{4\pi} \left[ \frac{R 2^{n+1}}{(X + z - x_{2})^{2} - (X_{2} + n L_{z} + (-1))^{n} a)^{2} + h^{2}} \right] \right\},$$

$$(1.2.11)$$

where  $z_{k11}$ ,  $z_{k12}$ ,  $z_{k21}$  and  $z_{k22}$  denote the minimum and maximum values of z at which reflected sounds for  $w_{k1}(z,n)$  and  $w_{k2}(z,n)$  can be received :

$$z_{k11} = -\frac{n L_{z} + (-1)^{n} a + X_{2}}{n L_{z} + (-1)^{n} a + L_{z} / 2 + b} X_{0} + X - X_{2}$$

$$z_{k12} = -\frac{n L_{z} + (-1)^{n} a + X_{2}}{n L_{z} + (-1)^{n} a + L_{z} / 2} X_{1} + X - X_{2},$$

-42-



Fig.1.2.5 Schematic analysis of sound reflections from the corner of the road intersection.

$$Z_{k21} = \frac{n L_{z} - (-1)^{n} a + X_{2}}{n L_{z} + (-1)^{n} a + L_{z}/2} X_{1} - X + X_{2},$$

$$Z_{k22} = -\frac{n L_{z} - (-1)^{n} a + X_{2}}{n L_{z} + (-1)^{n} a + L_{z}/2} X_{0} - X$$

$$+ X_{2}.$$
(1.2.12)

The objective total sound propagation characteristic w(z) is obtained by summing up each sound propagation characteristic,  $w_d(z)$ ,  $w_h(z)$  and  $w_k(z)$  (the peak value of w(z) is normalized in advance).

## 2.3 Experimental consideration

In the experimental considerations, the passing vehicles are classified into three categories, i.e., heavy vehicles, light vehicles and motorcycles. Table 1.2.1 shows the number of vehicles in each lane passing directly in front of the observation point during the measurement time interval of 30 minutes. The values of moment statistics,  $\langle Y^{m} \rangle$ , were obtained by averaging peak values of the observed elementary response wave forms of the noise intensities generated from each type vehicle. Figure 1.2.6 shows an example of the estimated sound propagation characteristic w(z) for the up-lane (the values of R<sub>1</sub> and R<sub>2</sub> are selected to 0.98). Based on the estimated sound propagation characteristics, Fig.1.2.7 shows a comparison between the theoretically predicted curves using the proposed method and the experimentally sampled points for the cumulative distribution form

	Up-Lane	Down-Lane
leavy vehicle	4	13
light vehicle	48	145
lotorcycle	7	15

Table 1.2.1 Number of vehicles passing directly in front of the

terval.

-30

-20

-10

observation point during the measurement time in-



Fig.1.2.6 An example of the estimated sound propagation characteristic (Up-Lane).

0

10

20

z (m)

30



Fig.1.2.7 A comparison between experimentally sampled points and theoretically predicted curves using the proposed method for the cumulative distribution form above the level value of 53 dB. Experimentally sampled points are marked by  $\bullet$  and theoretically predicted curves are respectively shown as ———, the first approximation of Eq.(1.2.4); ———, the second approximation (and the third approximation).

of the noise level fluctuation L. With consideration to the existence of background noises, a conditional cumulative distribution function with a restricted range above 53 dB has been predicted.<sup>(8)</sup> Moreover, by introducing the convolution integral for two statistically independent noise intensities as another method of considering the existence of background noises, a comparison between the theoretically predicted curve and the experimentally sampled points is shown in Fig.1.2.8. From these figures, the theoretically predicted results agree well with the experimentally observed values.

2.4 Conclusion

In this chapter, a general method for prediction of road traffic noise generated from the arbitrary traffic flow passing through a Ttype road intersection with sound insulation barriers on both sides of the road based on theoretical estimation of the sound propagation characteristic with the introduction of a well-known image method has been proposed. The effectiveness of the proposed prediction method has been experimentally confirmed by applying it to observed road traffic noise data.

This research is still in the early stages and the work reported here has focussed mainly on its methodological aspects. Accordingly, problems remain, e.g., application of the proposed method to the other actual engineering situations, such as other types of road intersections and/or the non-stationary traffic flow, etc.



Fig.1.2.8 A comparison between experimentally sampled points and theoretically predicted curve using the convolution integral for the cumulative distribution form of the noise level distribution. Experimentally sampled points are marked by • and theoretically predicted curve is shown as Chapter 3 A Statistical Method for Prediction of the Sound Insulation Effect of Barriers

## 3.1 Introduction

Sound insulation barriers are often constructed to produce attenuation of environmental noise such as the road traffic noise discussed in the previous chapters. The acoustical design and/or evaluation problems of such typical noise control systems have already been considered by many researchers. (17)-(25) Almost all these studies have been confined to the deterministic or average evaluation of the shielding effects. In an actual noise environment, however, the noise fluctuation emitted from sound sources often shows an irregular time pattern with intricate ups and downs, and various probability distribution forms other than the well-known Gaussian distribution form. Furthermore, statistics such as median,  $L_x$  noise level and  $L_{eq}$  evaluation index are of considerable importance for actual noise evaluation and regulation problems. It is necessary therefore that an explicit expression of the output noise level or noise intensity distribution form be established in close relation to the frequency characteristic of the input random noise fluctuation and the sound insulation barrier.

From the methodological point of view, there are two approaches for prediction of the above probability distribution. One approach would be construction of a unified prediction method by reinforcement

-49-

of the existing sound insulation barrier, with due consideration of acoustic characteristic change. An alternative approach would be consideration of the resultant probability distribution due to a newlyconstructed sound insulation barrier, rather than reinforcing or changing the existing system. In this chapter, a simplified and unified statistical evaluation method for predicting output noise level (or noise intensity) distribution based on the latter approach is theoretically proposed. The effectiveness of the proposed prediction method has been experimentally confirmed by applying it to actually observed noise data.

3.2 Theoretical consideration

3.2.1 General expression for noise intensity distribution

In order to predict the noise evaluation index,  $L_x$ , to find a general explicit expression of the probability density function with respect to the noise level or noise intensity fluctuations must first be found. As shown in Chapter 1, the generally applicable expression on the above-mentioned probability density function for noise intensity fluctuation is of a series expansion type taking a Gamma distribution as the first expansion term. More specifically, paying special attention to the fact that the noise intensity, x, always fluctuates in a non-negative region  $[0,\infty)$ , the probability density function P(x)can be generally expressed in the form of a statistical Laguerre

-50-

series expansion<sup>(13)</sup>:

P (x) = P<sub>r</sub> (x; m, s) 
$$\left[ 1 + \sum_{n=3}^{\infty} b'_{n} L_{n}^{(m-1)} \left( \frac{x}{s} \right) \right]$$
 (1.3.1)

with

$$m = \frac{\langle x \rangle^{2}}{\langle (x - \langle x \rangle)^{2} \rangle},$$
  

$$s = \frac{\langle (x - \langle x \rangle)^{2} \rangle}{\langle x \rangle},$$
(1.3.2)

$$b'_{n} = \frac{\Gamma (m) n !}{\Gamma (m+n)} \left\langle L_{n}^{(m+1)} \left( \frac{X}{s} \right) \right\rangle$$
$$= \sum_{k=0}^{n} \frac{(-1)^{k}}{k !} \frac{\Gamma (m) n !}{\Gamma (n-k+1) \Gamma (m+k)} \frac{\langle X^{k} \rangle}{s^{k}}$$
(1.3.3)

and

$$P_{\Gamma}(x; m, s) = \frac{1}{\Gamma(m) s^{m}} e^{-x/s} x^{m-1}.$$
(1.3.4)

(Gamma distribution)

At this time, the cumulative distribution function,  $Q(x)(\Delta f_0^X P(\zeta) d\zeta)$ , for the noise intensity fluctuation, which is very important for the purpose of finding any  $L_X$  noise level, is expressed as

$$Q(x) = \int_{0}^{x/s} P_{r}(\xi ; m) d\xi + P_{r}\left(\frac{x}{s} ; m+1\right) \sum_{n=3}^{\infty} A_{n}L_{n-1}(m)\left(\frac{x}{s}\right)$$
(1.3.5)

with

$$P_{\Gamma}(\xi;m) \triangleq \frac{1}{\Gamma(m)} e^{-\xi \xi m - 1}$$
(1.3.6)

(standard Gamma distribution)

and

$$A_{n} \triangleq \frac{(n-1)!\Gamma(m+1)}{\Gamma(m+n)} \left\langle L_{n}^{(m-1)} \left(\frac{x}{s}\right) \right\rangle$$
$$= \sum_{k=0}^{n} \frac{(-1)^{k}}{k!} \frac{(n-1)!\Gamma(m+1)}{\Gamma(n-k+1)\Gamma(m+k)} \frac{\langle x^{k} \rangle}{s^{k}} .$$
(1.3.7)

3.2.2 Relationship between frequency characteristics of a barrier and parameters contained in the cumulative distribution function

Let  $X_i$  (i=1,2,...,N) be the input noise intensity fluctuation existing in the ith frequency band-width of the noise fluctuation wave emitted from the sound source. And let the transfer coefficient  $a_i$ (i=1,2,...,N) reflect the intensity frequency characteristic of the sound insulation barrier at the center frequency,  $f_{ci}$ , of the ith octave band (or one-third-octave band). From the additive property of energy quantities, the over-all noise intensity, x, attenuated by the sound insulation barrier at an observation point is given explicitly by

$$\mathbf{x} = \sum_{i=1}^{N} \mathbf{a}_{i} \mathbf{X}_{i}. \tag{1.3.8}$$

Based on information on the statistical property of  $X_i$  and the intensity frequency characteristic  $a_i$  (i=1,2,...,N), the pth order moment,  $\langle x^p \rangle$ , for x can be easily estimated as

$$\langle \mathbf{x}^{\mathbf{p}} \rangle = \sum_{\mathbf{i}+\mathbf{j}+\ldots+\mathbf{n}=\mathbf{p}} \frac{\mathbf{p}^{\mathbf{j}}}{\mathbf{i} \mathbf{j} \mathbf{j} \mathbf{j} \mathbf{k} \ldots \mathbf{n} \mathbf{l}}$$

$$\cdot \mathbf{a}_{\mathbf{1}}^{\mathbf{i}} \mathbf{a}_{\mathbf{2}}^{\mathbf{j}} \ldots \mathbf{a}_{\mathbf{N}}^{\mathbf{n}} \langle \mathbf{X}_{\mathbf{1}}^{\mathbf{i}} \mathbf{X}_{\mathbf{2}}^{\mathbf{j}} \ldots \mathbf{X}_{\mathbf{N}}^{\mathbf{n}} \rangle .$$

$$(1.3.9)$$

Therefore, based on these lower and higher order moments, many values of parameters m, s and  $A_n$  can be calculated by use of Eqs.(1.3.2) and (1.3.7). The cumulative distribution function of the output noise intensity fluctuation can be evaluated by use of Eq.(1.3.5). The effect of the intensity frequency characteristic and the moment statistics of the incident random noise on the cumulative distribution function is explicitly reflected in each parameter m, s and  $A_n$ .

# 3.2.3 Estimation of the frequency characteristic of a sound insulation barrier

Let us consider the sound insulation barrier as shown in Fig.1.3. 1. Assuming that the fixed sound source can be regarded as a point source and that the barrier is sufficiently long in the horizontal



Fig.1.3.1 Location of sound insulation barrier.



Fig.1.3.2 Layout of sound source, barrier and two observation points. direction, the Fresnel number,  $N_i$ , at the center frequency,  $f_{ci}$ , of the ith octave band (or one-third-octave band) can be determined as

$$N_1 = 2 \delta f_{c1} / C$$
,  $\delta = d_1 + d_2 - d$ . (1.3.10)

where C is the speed of sound. Based on the value of  $N_i$ , after calculating the sound attenuation,  $\Delta L_i$ , of the barrier on Maekawa's acoustical evaluation chart<sup>(17)</sup> for barriers, the intensity frequency characteristic is explicitly evaluated by

$$a_i = 1 \ 0^{-\Delta L_i/10} \quad (i = 1, 2, ..., N)$$
. (1.3.11)

In this case,  $X_i$  (i=1,2,...,N) in Eq.(1.3.8) is the noise intensity fluctuation existing in the ith frequency band-width of the observed noise intensity fluctuation at the observation point 0 (see Fig.1.3. 1), before the barrier is constructed.

### 3.3. Experimental consideration

The experiment was done at night (20.00 P.M.-03.00 A.M.) in a playground to avoid the effect of surrounding background noises. The layout of the sound source, the barrier and two observation points is shown in Fig.1.3.2. A barrier made of plywood panel (height : 1.79m, width : 6.32m, thickness : 13.5mm) was used.

A road traffic noise wave was used as one example of actual envi-

ronmental random noise. Using two sound level meters, two kinds of A-weighted over-all noise intensity fluctuation waves, x and x', were recorded on a data recorder before and after construction of the barrier. For simplification of experimental procedures, the noise intensity fluctuation wave was observed by use of the octave band analysis.

The intensity frequency characteristic,  $a_i$ , of the barrier estimated using Maekawa's evaluation chart for a barrier is shown in Table 1.3.1. The effect of reflection from the ground has been taken into consideration.

Figure 1.3.3 shows a comparison between the theoretically predicted curves using the proposed method and the experimentally sampled points obtained from the observed data x, for the cumulative distribution form of the noise intensity fluctuation in a case in which the height of the observation point was 0.8 m. In another case in which the height of the observation point was 1.3 m, a comparison between theory and experiment is shown in Fig.1.3.4.

In each case, it can be seen that the theoretical curves come closer to the experimentally sampled points as the number of correction terms increases.

### 3.4 Conclusion

In this chapter, a statistical prediction method for output random noise fluctuation has been proposed in a unified type of

of octave band (Hz)	h=0.8(m)	h=1.3(m)
250	0.07771	0.19498
500	0.04023	0.05370
1000	0.01943	0.04897
2000	0.00996	0.02754
4000	0.00479	0.01479

Table 1.3.1 The estimated values of intensity frequency characteristic, a<sub>1</sub>, for a sound insulation barrier.



Fig.1.3.3 A comparison between experimentally sampled points and theoretically predicted curves using the proposed method for the cumulative distribution form of the noise intensity fluctuation (height of observation point : 0.8 m). Experimentally sampled points are marked by  $\bullet$  and theoretically predicted curves are respectively shown as \_\_\_\_\_, the first approximation of Eq.(1.3.5); ----, the second approximation; \_\_\_\_, the third approximation.



Fig.1.3.4 A comparison between experimentally sampled points and theoretically predicted curves using the proposed method for the cumulative distribution form of the noise intensity fluctuation (height of observation point : 1.3 m). Experimentally sampled points are marked by  $\bullet$  and theoretically predicted curves are respectively shown as \_\_\_\_\_, the first approximation of Eq.(1.3.5); ----, the second approximation; \_\_\_\_\_, the third approximation.

probability distribution form, when a general distribution type stationary random noise has been attenuated by a sound insulation barrier.

Research on the statistical evaluation of sound insulation barriers is still in the early stages. This study has only focussed on some fundamental aspects. Many problems still remain for application to actual case which will be the subject of future studies. PART II

1

MUTUAL RELATIONSHIP BETWEEN VARIOUS TYPES OF NOISE EVALUATION INDICES CONNECTED WITH  $\rm L_{x}$  AND  $\rm L_{eq}$ 

Chapter 1 A Method for Estimation of  $\rm L_{eq}$  Noise Evaluation Index by Use of Moment Statistics

1.1 Introduction

The following sampling method is often employed with the aid of a digital measurement technique to evaluate an  $L_{eq}$  noise evaluation index :

$$L_{eq} = 1 \ 0 \ 1 \ 0 \ g_{10} \frac{1}{N} \sum_{i=1}^{N} 1 \ 0 \ L_{i/10} \ , \qquad (2.1.1)$$

where N denotes the total number of the noise level  $L_i$  (i=1,2,...,N) sampled over a measurement time interval. Many instruments employ this evaluation method by using a usual digital sound level meter with quantized levels. From the viewpoint of signal processing, however, the following fundamental problems remain :

- (1) It is necessary to obtain many level data with a fairly fine sampling period, since the noise intensity fluctuation after the anti-logarithmic transformation of the sampled level datum :
  - $x_i = x_0 \cdot 1 0^{Li/10}$  (x<sub>0</sub> : reference noise intensity) (2.1.2)

fluctuates with large excursions, as compared with the original decibel-scaled level fluctuation.

(2) In principle, the mean value of the noise intensity fluctuation

should be given by the sample mean operation based on the noise intensity data especially with an equally quantized intensity amplitude. If one uses the noise intensity fluctuation after transforming the measured noise level fluctuation with an equally quantized level amplitude into the intensity scale through the anti-logarithmic transformation, a calculation error of the intensity mean will occur because of the above level quantization.

When the noise evaluation index,  $L_{eq}$ , is calculated by the usual method (see Eq.(2.1.1)) based on its original definition with the use of actually measured data with digitalization, one can not avoid essentially the above two fundamental problems. Since this evaluation index is originally in the form of statistical information, a probabilistic relationship between this Leg and the reliable and stable statistical information on noise level fluctuation should exist. If this relationship can be found theoretically and the objective Leg from the statistical information on a decibel scale can be obtained by this theory, the above-mentioned fundamental problems can be avoided. That is, it is convenient to utilize explicitly the statistical information on the decibel-scaled level fluctuation itself with the aid of the theory for evaluating Leg, since the above information is fairly stable and reliable, being based on the averaging operation supported by a large amount of data. In this case, it would seem to be appropriate to derive first a general estimation method by using a unified expression on the noise level distribu-Furthermore, the actual environmental noise level fluctuation.

-61-
tions encountered in daily life exhibit various types of probability distributions, due to the diversified types of noise fluctuation.

In consideration of these practical points, a general explicit expression for estimating  $L_{eq}$  using the moment statistics of the noise level fluctuation is proposed in this chapter after introduction of a general explicit expression on the arbitrary type noise level distribution form. More specifically, an expression for an arbitrary non-linear type expectation value is first derived based on an approach equivalent to that used for the well-known Gaussian distribution (this approach is quite similar to that discussed in Chapter 1, Part I). Next, an estimation method for the objective  $L_{eq}$  based on this expression is newly derived in a hierarchical form containing a well-known simplified estimation method derived under the assumption of ordinary Gaussian distribution as a special case.<sup>(27),(28)</sup>

Finally, the effectiveness of the proposed estimation method has been experimentally confirmed by applying it to actually observed road traffic noise data.

1.2 Theoretical consideration

1.2.1 General theory for estimation of mean value of arbitrary nonlinear function type random variables To begin the analysis, let us introduce a statistical Hermite series expansion<sup>(13)</sup> as a probability density function universally applicable to arbitrary non-Gaussian distribution forms of the noise level fluctuation L, as follows :

P (L) = N (L; 
$$\mu_{L}, \sigma_{L}^{2}$$
)  $\left\{ 1 + \sum_{n=3}^{\infty} B_{n} H_{n} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) \right\}$  (2.1.3)

with

$$\mu_{L} \triangleq \langle L \rangle , \qquad \sigma_{L}^{2} \triangleq \langle (L - \mu_{L})^{-2} \rangle ,$$

$$B_{n} \triangleq \frac{1}{n!} \left\langle H_{n} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) \right\rangle , \qquad (2.1.4)$$

where N(L ;  $\mu_L$ ,  $\sigma_L^2$ ) denotes the well-known Gaussian distribution with the mean  $\mu_L$  and the variance  $\sigma_L^2$ :

N (L; 
$$\mu_{\rm L}, \sigma_{\rm L}^2$$
)  $\triangleq \frac{1}{\sqrt{2 \pi \sigma_{\rm L}}} \exp \left[ (L - \mu_{\rm L})^2 / 2 \sigma_{\rm L}^2 \right].$  (2.1.5)

By using the relationship between the Gaussian distribution and the Hermite polynomial  $^{(13)}$  :

N (L; 
$$\mu_L$$
,  $\sigma_L^2$ )  $H_n\left(\frac{L-\mu_L}{\sigma_L}\right) = (-1)^n \sigma_L^n$   
 $\cdot \frac{d^n}{dL^n}$  N (L;  $\mu_L$ ,  $\sigma_L^2$ ), (2.1.6)

Eq.(2.1.3) can be rewritten as

$$P(L) = N(L; \mu_{L}, \sigma_{L}^{2}) + \sum_{n=3}^{\infty} B_{n}(-1)^{n} \sigma_{L}^{n} \frac{d^{n}}{dL^{n}} N(L; \mu_{L}, \sigma_{L}^{2}).$$
(2.1.7)

Based on the above relationship, after introducing an arbitrary nonlinear function g(.) with respect to L, its expectation value is given by

$$\langle g(L) \rangle \triangleq \int_{-\infty}^{\infty} g(L) P(L) dL = \int_{-\infty}^{\infty} g(L) N(L; \mu_{L}, \sigma_{L}^{2}) dL + \sum_{n=3}^{\infty} B_{n}(-1)^{n} \sigma_{L}^{n} \int_{-\infty}^{\infty} g(L) \cdot \frac{d^{n}}{dL^{n}} N(L; \mu_{L}, \sigma_{L}^{2}) dL .$$

$$(2.1.8)$$

Furthermore, by introducing the following notation :

$$\langle g(L) \rangle_{\text{Gauss}} \triangleq \int_{-\infty}^{\infty} g(L) N(L; \mu_L, \sigma_L^2) dL$$
 (2.1.9)

and carrying out the n-time integration by parts in the second term of Eq.(2.1.8), the following relationship can be obtained :

$$\langle g (L) \rangle = \langle g (L) \rangle_{Gauss} + \sum_{n=3}^{\infty} B_n \sigma_L^n \left\langle \frac{d^n}{dL^n} g (L) \right\rangle_{Gauss}$$
 (2.1.10)

The probabilistic boundary condition due to the reasonable limiting property given by

$$\frac{d^{-1}}{d L^{n}} N (L; \mu_{L}, \sigma_{L}^{2}) \bigg|_{L=-\infty \text{ or } L=\infty} = 0 \qquad (2.1.11)$$

has been used for deriving the above relationship.

By using Eq.(2.1.10), the expectation value  $\langle g(L) \rangle$  in terms of an arbitrary non-Gaussian distribution can be equivalently calculated based on an expectation operation in terms of an ordinary Gaussian distribution. The advantage of using Eq.(2.1.10) is that the wellknown mathematical properties for Gaussian distribution can be utilized, since this probability distribution function plays an important role in the field of statistics and many related properties have already been investigated.

1.2.2 Estimation method of Leg noise evaluation index

Consider an Leg noise evaluation index defined as

$$L_{eq} \left( \triangleq 1 \ 0 \ 1 \ 0 \ g_{10} \frac{\langle x \rangle}{x_0} \right) = 1 \ 0 \ 1 \ 0 \ g_{10} \langle 1 \ 0 \ ^{L/10} \rangle .$$
(2.1.12)

After employing the following function form :

$$g(L) = 1 0^{L/10}$$
(2.1.13)

as a non-linear function g(x) in Eq.(2.1.10), the following relationship can be derived by using the expectation operation in terms of a well-known Gaussian distribution :

$$(g(L)) = Gauss \triangleq \int_{-\infty}^{\infty} 1 \ 0^{L/10} \frac{1}{\sqrt{2 \pi \sigma_L}} e x p \left[ -\frac{(L - \mu_L)^2}{2 \sigma_L^2} \right] dL$$
  
=  $e x p (\mu_L / M + \sigma_L^2 / 2 M^2) (M \triangleq 1 \ 0 / 1 \ n \ 1 \ 0)$ 

(2.1.14)

$$\left\langle \frac{d^{n}}{dL^{n}}g(L)\right\rangle = e \times p (\mu_{L}/M + \sigma_{L}^{2}/2M^{2})/M^{n}. (2.1.15)$$
Gauss

Through substitution of Eqs.(2.1.14) and (2.1.15) into Eq.(2.1.10), the expectation value of the non-linear function, g(x), based on the expectation operation in terms of an arbitrary non-Gaussian distribution can be expressed in a hierarchical form, as follows :

$$\langle g (L) \rangle = e \times p (\mu_{L} / M + \sigma_{L}^{2} / 2 M^{2})$$
  
 $\cdot \left[ 1 + \sum_{n=3}^{\infty} B_{n} \sigma_{L}^{n} / M^{n} \right].$ 
(2.1.16)

Since the expansion coefficient  $B_n$  (n=1,2,...) is expressed in a specific form using its definition, as follows :

$$B_{3} = \frac{1}{3!} \left\langle H_{3} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) \right\rangle = \frac{1}{6 \sigma_{L}^{3}} \left\langle (L - \mu_{L})^{3} \right\rangle,$$

$$B_{4} = \frac{1}{4!} \left\langle H_{4} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) \right\rangle = \frac{1}{2 4 \sigma_{L}^{4}} \left[ \left\langle (L - \mu_{L})^{4} \right\rangle - 3 \sigma_{L}^{4} \right],$$

$$B_{5} = \frac{1}{5!} \left\langle H_{5} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) \right\rangle = \frac{1}{1 2 0 \sigma_{L}^{5}} \left\langle (L - \mu_{L})^{5} \right\rangle, \quad \dots, \quad (2.1.17)$$

Eq.(2.1.16) can consequently be expressed as

$$\langle g (L) \rangle = e \times p (\mu_{L} / M + \sigma_{L}^{2} / 2 M^{2}) \left[ 1 + \frac{1}{6 M^{3}} \langle (L - \mu_{L})^{3} \rangle + \frac{1}{2 4 M^{4}} (\langle (L - \mu_{L})^{4} \rangle - 3 \sigma_{L}^{4} \rangle + \cdots \right].$$
 (2.1.18)

After substitution of Eq.(2.1.18) into Eqs.(2.1.12) and (2.1.13), the objective explicit expression of hierarchically estimating  $L_{eq}$  can be obtained as

$$L_{eq} = \mu_{L} + \sigma_{L}^{2} / 2 M + M \ln \left[ 1 + \frac{1}{6 M^{3}} \langle (L - \mu_{L})^{3} \rangle + \frac{1}{2 4 M^{4}} (\langle (L - \mu_{L})^{4} \rangle - 3 \sigma_{L}^{4}) + \cdots \right].$$
(2.1.19)

That is, from Eq.(2.1.19),  $L_{eq}$  can be given by the lower and higher order moment statistics for the noise level fluctuation of an arbitrary non-Gaussian distribution type. It should be noticed that the above estimation formula agrees completely with a well-known simplified estimation formula, derived under assumption of an ordinary Gaussian distribution as the first approximation (27),(28):

$$L_{eq} = \mu_{L} + 0.115 \sigma_{L}^{2}, \qquad (2.1.20)$$

since the higher order expansion terms become zero for this case.

### 1.3 Experimental consideration

For the purpose of confirming the effectiveness of the proposed estimation method, two kinds of road traffic noise data measured using a digital sound level meter were employed. The first was measured near a national main road in a large city. At that time, the volume of traffic was 244 vehicles per 10 minutes. The second was measured near a country road in a rural district with a traffic volume of 19 vehicles per 10 minutes. The time interval for measurement and its sampling period were selected to be 10 minutes and 1 second, respectively. In addition, the values of  ${\rm L}_{\rm eq}$  in each case were simultaneously measured by a precision integrating sound level meter for L<sub>eq</sub>. To simplify the notation of these cases, let us define the former case as " Case A " and the latter as " Case B ". Figure 2.1.1 shows the arrangements for measuring road traffic noise at an observation point for Case A. Figure 2.1.2 shows the arrangements for measuring road traffic noise at an observation point for Case B.

Figure 2.1.3 shows a comparison between the theoretically estimated curve using only the Gaussian distribution and the experimentally sampled points in the cumulative distribution form of the noise level fluctuation for Case A. From this figure, the theoretically estimated curve using only the Gaussian distribution agrees approximately with the experimentally sampled points. Thus, the data for Case A exhibits approximately the Gaussian distribution form. More-



Fig.2.1.1 Arrangements for measuring road traffic noise at an observation point (Case A).



Fig.2.1.2 Arrangements for measuring road traffic noise at an observation point (Case B).



Fig.2.1.3 A comparison between experimentally sampled points and theoretically estimated curve using Eq.(2.1.5) for the cumulative distribution form of the noise level fluctuation (Case A). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curve is shown as

over, a comparison between the theoretically estimated curve using only the Gaussian distribution and the experimentally sampled points for Case B is shown in Fig.2.1.4. From this figure, the theoretically estimated curve using only the Gaussian distribution does not agree well with the experimentally sampled points. The data for Case B therefore strongly exhibits the non-Gaussian distribution form. After applying the proposed estimation method to these two kinds of actual road traffic noise data, the estimated results of  $L_{eq}$  can be obtained.

Table 2.1.1 shows the estimated results of Leg for Case A using the proposed estimation method. The expansion expression of Eq.(2.1. 19) from the first expansion term corresponding to the wellknown simplified estimation method, Eq.(2.1.20), is defined to the nth order expansion term, as the (n-1)th approximation of the estimated Leg. According to these estimated results, the estimated value, using a well-known simplified estimation method derived under the assumption of an ordinary Gaussian distribution, agrees approximately with the measured value, since this data exhibits approximately the Gaussian distribution. More precisely, however, the estimated values using the proposed method tend agree well with the successive addition of higher order expansion terms. Table 2.1.2 shows the estimated results for Case B. According to these estimated results, the estimation accuracy using the well-known simplified estimation method is not sufficient for evaluation, since this data exhibits strongly the non-Gaussian distribution form. On the other hand, it is clear that the successive addition of higher order



Fig.2.1.4 A comparison between experimentally sampled points and theoretically estimated curve using Eq.(2.1.5) for the cumulative distribution form of the noise level fluctuation (Case B). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curve is shown as

Measured value of Leg (dB)	Estimated values using the proposed method (dB)
86.0	86.3 (the first approximation)
	86.2 (the second approximation)
	86.2 (the third approximation)
	86.1 (the fourth approximation)

Table 2.1.1 The estimated results of Leg using the proposed method (CaseA).

Measured value	Estimated values using the
of Leg (dB)	proposed method (dB)
76.4	72.9(the first approximation)
	77.8(the second approximation)

. .

0

~

TT L

...

.

. .

expansion terms moves the values estimated theoretically using the proposed method closer to the value measured experimentally. In each case, the estimated results agree well with the experimental results.

### 1.4 Conclusion

In this chapter, a new trial for estimating an  $L_{eq}$  noise evaluation index using the moment statistics of the noise level fluctuation has been proposed. More specifically, an explicit expression for an arbitrary non-linear function type expectation value which is equivalent to that for an ordinary Gaussian distribution has been first derived. Next, based on this convenient expression, a new trial of estimating  $L_{eq}$  using the moment statistics of the noise level fluctuation has been proposed, especially in a hierarchical form. Finally, the effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data.

This study is in the early stages and has therefore focussed only on the introduction of some methodological aspects. Accordingly, many future problems still remain. First, this method must be applied to many other actual cases to broaden and confirm its practical effectiveness. In addition, based on this method, a specific on-line measurement system for the sequential measurement of  $L_{eq}$  should be constructed for practical use.

# Chapter 2 Mutual Relationship between Several Noise Evaluation Indices Connected with $\rm L_{x}$ and $\rm L_{eq}$

### 2.1 Introduction

As is well-known, two noise evaluation indices,  $L_x$  and  $L_{eq}$ , play an important role in the field of noise evaluation and regulation problems. It is very important to find a general theory for the mutual relationships between several noise evaluation indices connected with  $L_x$  and  $L_{eq}$ , in order to grasp the systematical evaluation of environmental random noise. Up to now, various studies for the mutual relationship of this kind have already been investigated by many researchers. <sup>(27)-(34)</sup> Almost all these studies, however, were confined only to practical methods derived by applying directly the well-known linear regression analysis method to actually observed data or approximate methods derived under the assumption of a well-known Gaussian distribution. If a universally and objectively more precise relationship between these statistical evaluation indices is to be found, these conventional methods can not be used.

In this chapter, a general theory for the mutual relationship between several types of noise evaluation indices in relation to two typical indices,  $L_x$  and  $L_{eq}$  is proposed. More specifically, a general method for estimation of  $L_{eq}$  using several specific  $L_x$  is proposed in a general form including a simplified estimation method derived under the assumption of an ordinary Gaussian distribution as a special case. Next, a new trial estimating the original noise level distribution using the measured values of  $L_{eq}$  and several specific  $L_x$  is proposed. The effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data.

### 2.2 Theoretical consideration

## 2.2.1 Method for estimation of ${\rm L}_{eq}$ noise evaluation index by use of several specific ${\rm L}_{\rm X}$

Let us introduce the moment generating function  $M_L(\theta)$  with respect to the noise level fluctuation  $L(\Delta 10\log_{10} x/x_0)$ , exhibiting an arbitrary non-Gaussian distribution form, as follows :

$$M_{L}(\theta) = \langle e x p (\theta M 1 n x / x_{0}) \rangle \quad (M \leq 10/\ln 10) . (2.2.1)$$

The mathematical relationship between the arbitrary order cumulant  $\kappa_{Ln}$  (n=1,2,...) with respect to L and the moment generating function  $M_L$  ( $\theta$ ) is given by

$$M_{L}(\theta) = e \times p\left(\sum_{n=1}^{\infty} \frac{\kappa_{Ln}}{n!} \theta^{n}\right). \qquad (2.2.2)$$

By replacing  $\theta$  with 1/M in Eqs.(2.2.1) and (2.2.2), the following relationship can easily be obtained as

$$\langle \mathbf{x} \rangle = \mathbf{x} \mathbf{o} \mathbf{e} \mathbf{x} \mathbf{p} \left( \sum_{n=1}^{\infty} \frac{1}{n!} \cdot \frac{\kappa_{Ln}}{M^{n}!} \right).$$
 (2.2.3)

A substitution of Eq.(2.2.3) into the definition of  $L_{eq}$  thus yields a general expression of a series expansion type for estimating  $L_{eq}$ , as follows :

$$L_{eq} \triangleq 1 \ 0 \ 1 \ 0 \ g_{10} \frac{\langle x \rangle}{x_{0}}$$

$$= \kappa_{L1} + \frac{\kappa_{L2}}{2 M} + \frac{\kappa_{L3}}{6 M^{2}} + \frac{\kappa_{L4}}{2 4 M^{3}} + \frac{\kappa_{L5}}{1 2 0 M^{4}} + \cdots$$

$$= \mu_{L} + 0 \ . \ 1 \ 1 \ 5 \ \sigma_{L}^{2} + 8 \ . \ 8 \ 4 \times 1 \ 0^{-3} \kappa_{L3}$$

$$+ 5 \ . \ 0 \ 9 \times 1 \ 0^{-4} \kappa_{L4} + 2 \ . \ 3 \ 4 \times 1 \ 0^{-5} \kappa_{L5} + \cdots \qquad (2.2.4)$$

When the noise level fluctuation exhibits an ordinary Gaussian distribution, Eq.(2.2.4) is reduced to a well-known simplified estimation method derived under the assumption of a well-known Gaussian distribution (27),(28):

$$L_{eg} = \mu_{L} + 0.115 \sigma_{L}^{2} = L_{50} + 0.115 \sigma_{L}^{2}, \qquad (2.2.5)$$

since the higher order cumulants  $\kappa_{\rm Ln}~(n=3,4,\ldots)$  in this case are equal to zero.

A general explicit expression of the cumulative distribution function for the noise level fluctuation L is expressed in a statistical Hermite series expansion,<sup>(13)</sup> as follows :

$$Q (L) = \int_{-\infty}^{L} N (\xi; \mu_{L}, \sigma_{L}^{2}) d\xi$$
  
-  $\frac{\kappa_{L3}}{6 \sigma_{L}^{2}} N (L; \mu_{L}, \sigma_{L}^{2}) H_{2} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right)$   
-  $\frac{\kappa_{L4}}{2 4 \sigma_{L}^{3}} N (L; \mu_{L}, \sigma_{L}^{2}) H_{3} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) - \cdots (2.2.6)$ 

According to the definition of  $\text{L}_{\rm X},$  the following relationship can easily be obtained :

$$1 - \frac{x}{1 \ 0 \ 0} = \int_{-\infty}^{Lx} N \ (\xi \ ; \ \mu \ L, \ \sigma \ L^2) \ d \ \xi$$
$$- \frac{\kappa \ L3}{6 \ \sigma \ L^2} N \ (L \ x \ ; \ \mu \ L, \ \sigma \ L^2) \ H \ 2 \left( \frac{L \ x - \ \mu \ L}{\sigma \ L} \right)$$
$$- \frac{\kappa \ L4}{2 \ 4 \ \sigma \ L^3} N \ (L \ x \ ; \ \mu \ L, \ \sigma \ L^2) \ H \ 3 \left( \frac{L \ x - \ \mu \ L}{\sigma \ L} \right) - \dots (2.2.7)$$

The application of N kinds of specific  $L_x$  ( $L_{x1}$ ,  $L_{x2}$ ,...,  $L_{xN}$ ) to Eq. (2.2.7) thus yields the N-dimensional simultaneous equations by regarding the cumulant statistics  $\kappa_{Ln}$ (n=3,4,...) as unknown parameters. From the practical point of view, the explicit expressions for estimating  $L_{eq}$  for cases with N=1 and N=2 are shown as

(1) N=1 (reference level 
$$L_{x1}$$
) :

$$L_{eg} = \mu_{L} + 0.115\sigma_{L}^{2} + 8.84 \times 10^{-3} \kappa_{L3}$$
(2.2.8)

with

$$\kappa_{L3} = \frac{6 \sigma_{L}^{2} F(Z_{1})}{N(Z_{1}; 0, 1) H_{2}(Z_{1})}, Z_{1} \triangleq \frac{L_{X1} - \mu_{L}}{\sigma_{L}}, \qquad (2.2.9)$$

where F(z) is defined as

F (Z) 
$$\triangleq \int_{-\infty}^{z} N(\xi; 0, 1) d\xi + \frac{X}{100} - 1.$$
 (2.2.10)

(2) N=2 (reference levels :  ${\rm L}_{x1}$  and  ${\rm L}_{x2})$  :

σL

$$L_{eq} = \mu_{L} + 0.115 \sigma_{L}^{2} + 8.84 \times 10^{-3} \kappa_{L3} + 5.09 \times 10^{-4} \kappa_{L4} \qquad (2.2.11)$$

with

$$\kappa_{L3} = \frac{6 \sigma_{L^2} [N(Z_2; 0, 1) H_3(Z_2) F(Z_1) - N(Z_1; 0, 1) H_3(Z_1) F(Z_2)]}{N(Z_1; 0, 1) N(Z_2; 0, 1) [H_2(Z_1) H_3(Z_2) - H_2(Z_2) H_3(Z_1)]},$$

$$\kappa_{L4} = \frac{24 \sigma_{L}^{3} [N(Z_{1}; 0, 1) H_{2}(Z_{1}) F(Z_{2}) - N(Z_{2}; 0, 1) H_{2}(Z_{2}) F(Z_{1})]}{N(Z_{1}; 0, 1) (Z_{2}; 0, 1) [H_{2}(Z_{1}) H_{3}(Z_{2}) - H_{2}(Z_{2}) H_{3}(Z_{1})]},$$

$$Z_{1} \triangleq \frac{L_{x1} - \mu_{L}}{Z_{2}} = \frac{L_{x2} - \mu_{L}}{Z_{2}} + \frac{L_{x2}$$

σL

2.2.2 Method for estimation of noise probability distribution by use

### of Leg and several specific ${\tt L}_{\tt X}$

In this section, let us consider a general method for estimation of the original noise level distribution by using the measured values of  $L_{eq}$  and several specific  $L_x$ . To begin the analysis, a statistical Laguerre series expansion<sup>(13)</sup> can be employed as a general expression of the noise intensity distribution. The cumulative distribution function of this expression is given after introducing a dimensionless variable,  $\eta(=x/s)$ , as follows :

$$Q(\eta) = \int_{0}^{\pi} P_{\Gamma}(\xi; m) d\xi$$

$$+ A'_{4} \frac{\eta^{m}}{24 (m)} e x p (-\eta) f_{2}(\eta) + \cdots$$
(2.2.13)

with

$$m = \mu_{\mathbf{x}}^2 / \sigma_{\mathbf{x}}^2, \quad s = \sigma_{\mathbf{x}}^2 / \mu_{\mathbf{x}},$$

$$A'_{n} \triangleq \frac{n ! \Gamma (m)}{\Gamma (m+n)} \left( L_{n}^{(m-1)} \left( \frac{x}{-} \right) \right) , \qquad (2.2.14)$$

where  $f_1(n)$  and  $f_2(n)$  are respectively defined as

$$f_{1}(\eta) \triangleq \eta^{2} - 2(m+2)\eta + (m+2)(m+1),$$

$$f_{2}(\eta) \triangleq -\eta^{3} + 3(m+3)\eta^{2} - 3(m+3)(m+2)\eta + (m+3)(m+2)(m+1).$$
(2.2.15)

Moreover,  $\mu_x$  and  $\sigma_x^2$  are the mean value of x and its variance, respectively. The distribution parameters, m and s, can be rewritten by using L<sub>eq</sub>, as follows :

 $(0.2 L e q + 2 1 o g_{10} x_{0} - 1 o g_{10} \sigma_{x}^{2})$ m = 1 0

$$(-0.1 \text{ Leq} - 1 \text{ og}_{10} \text{ x}_{0} + 1 \text{ og}_{10} \sigma_{x}^{2})$$
 (2.2.16)  
s = 1 0

From Eq.(2.2.13) and the definition of  ${\rm L}_{\rm X},~$  the following relationship can easily be obtained :

$$1 - \frac{X}{1 \ 0 \ 0} = \int_{0}^{\eta} \frac{x}{P_{r}} (\xi ; m) \ d\xi$$

+ A'  $\frac{\eta \mathbf{x}^{\mathbf{m}}}{6 \Gamma (\mathbf{m})}$  exp (-  $\eta \mathbf{x}$ ) f  $_{1} (\eta \mathbf{x})$ 

 $+ A'_{4} \frac{\eta_{x}^{m}}{2 4 (m)} e x p (-\eta_{x}) f_{2} (\eta_{x}) + \cdots,$ 

$$\eta_{\mathbf{x}} \triangleq \frac{\mathbf{X} \mathbf{o}}{1 \mathbf{0} \mathbf{1} \mathbf{x} / \mathbf{10}} .$$
(2.2.17)

Thus, the application of N kinds of specific  $L_x$  ( $L_{x1}$ ,  $L_{x2}$ ,...,  $L_{xN}$ ) to Eq.(2.2.17) yields N-dimensional simultaneous equations, by regarding the expansion coefficients,  $A'_n$  (n=3,4,...,N+2), as unknown parameters. After estimating the distribution parameters, m, s and  $A'_n$ , using the known values of  $L_{eq}$  and several specific  $L_x$ , one can evaluate the original noise level distribution by substituting the estimated values of  $A'_n$  into Eq.(2.2.13). At this time, the cumulative distribution function Q(L) with respect to L can be evaluated using the probability measure preserving transformation. From the practical point of view, the explicit expressions for estimating  $A'_n$  for cases with N=1 and N=2 are shown as

(1) N=1 (reference level :  $L_{x1}$ ) :

$$A'_{3} = \frac{6 \Gamma (m) G (x_{1})}{\eta_{1}^{n} f_{1} (\eta_{1}) e x p (-\eta_{1})}$$
(2.2.18)

with

$$\eta_{1} \triangleq \frac{X_{0}}{10^{1}} (2.2.19)$$

where  $G(\mathbf{x})$  is defined as

$$G(x) \triangleq 1 - \frac{x}{100} - \int_0^{\pi} P_{\Gamma}(\xi; m) d\xi.$$
 (2.2.20)

(2) N=2 (reference levels :  $L_{x1}$  and  $L_{x2}$ )

$$A'_{3} = \frac{6\Gamma(m) \{ \eta_{2}^{m} f_{2}(\eta_{2}) \exp(-\eta_{2}) G(x_{1}) - \eta_{1}^{m} f_{2}(\eta_{1}) \exp(-\eta_{1}) G(x_{2}) \}}{\eta_{1}^{m} \eta_{2}^{m} \exp\{-(\eta_{1} + \eta_{2})\} [f_{1}(\eta_{1}) f_{2}(\eta_{2}) - f_{1}(\eta_{2}) f_{2}(\eta_{1})]}$$

$$A'_{4} = \frac{24\Gamma(m) \{ \eta_{1}^{m} f_{1}(\eta_{1}) \exp(-\eta_{1}) G(x_{2}) - \eta_{2}^{m} f_{2}(\eta_{2}) \exp(-\eta_{2}) G(x_{1}) \}}{\eta_{1}^{m} \eta_{2}^{m} \exp\{ -(\eta_{1}+\eta_{2}) \} [f_{1}(\eta_{1}) f_{2}(\eta_{2}) - f_{1}(\eta_{2}) f_{2}(\eta_{1}) ]},$$

(2.2.21)

with

$$7_{1} \triangleq \frac{X_{0}}{10^{L \times 1/10}}, \qquad 7_{2} \triangleq \frac{X_{0}}{10^{L \times 2/10}}.$$

2.3 Experimental consideration

For the purpose of confirming the effectiveness of the proposed estimation method, two kinds of road traffic noise data have been used. Road traffic noise was measured near a national main road in a large city, where the traffic volume was more than 3000 vehicles per hour, and near a national main road in a rural district with a traffic volume of about 1000 vehicles per hour. The measurement time interval and its sampling period were selected to be 1 hour and 1 second, respectively. To simplify the notation of these cases, let us define the former case as " Case A " and the latter as " Case B ". Figure 2.2.1 shows the arrangements for measuring road traffic noise at an observation point for Case A, and Fig. 2.2.2 shows the arrangements for measuring road traffic noise at an observation point for Case B.

Figure 2.2.3 shows a comparison between the theoretically estimated curves using the statistical Hermite series expansion, Eq.(2.2. 6), and the experimentally sampled points in the cumulative distribution form of the noise level fluctuation for Case A. From this figure, the theoretically estimated curve using only the first expansion term (i.e., Gaussian distribution) agrees approximately with the experimentally sampled points. Thus, the data for Case A exhibits approximately the Gaussian distribution form. Figure 2.2.4 shows a comparison between the theoretically estimated curves and the experimentally sampled points for Case B. From this figure, the theoretically estimated curve using only the first expansion term (i.e., Gaussian distribution) does not agree well with the experimentally sampled points. Thus, the data for Case B exhibits the non-Gaussian distribution form. Figure 2. 2.5 shows a comparison between the theoretically estimated curves using the statistical Laguerre series expansion and the experimentally sampled points in the cumulative distribution form for the noise intensity fluctuation for Case A. From this figure, the theoretically estimated curve using only the first expansion term (i.e., Gamma distribution) does not agree well with the experimentally sampled points. The data for Case A exhibits the non-Gamma distribution form. A comparison between the theoretically estimated curves and the experimen-



Fig.2.2.1 Arrangements for measuring road traffic noise at an observation point (Case A).







Fig.2.2.3 A comparison between experimentally sampled points and theoretically estimated curves using Eq.(2.2.6) for the cumulative distribution form of the noise level fluctuation (Case A). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.2.6); ----, the second approximation (and the third approximation).



Fig.2.2.4 A comparison between experimentally sampled points and theoretically estimated curves using Eq.(2.2.6) for the cumulative distribution form of the noise level fluctuation (Case B). Experimentally sampled points are marked by • and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.2. 6); ----, the second approximation (and the third approximation.



Fig.2.2.5 A comparison between experimentally sampled points and theoretically estimated curves using Eq.(2.2.13) for the cumulative distribution form of the noise intensity fluctuation (Case A). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.2.13);----, the second approximation;----, the third approximation.

tally sampled points for Case B is shown in Fig.2.2.6. The theoretically estimated curve using only the first expansion term (i.e., Gamma distribution) agrees approximately with the experimentally sampled points. The data for Case B exhibits approximately the Gamma distribution.

Table 2.2.1 shows the estimated results of  $L_{eq}$  for Case A using the estimation method proposed in Section 2.2.1. According to these estimated results, the estimated value, using a well-known simplified estimation method derived under assumption of an ordinary Gaussian distribution, agrees approximately with the measured value, since this data exhibits approximately the Gaussian distribution form. More precisely, however, the estimated values using the proposed method tend to agree well with the successive addition due to other reference levels of  $L_X$ . Table 2.2.2 shows the estimated results for Case B. According to these estimated results, the accuracy of the well-known simplified estimation method is not sufficient for evaluation, since this data exhibits the non-Gaussian distribution form. The values estimated theoretically using the proposed method agree well with the measured value of  $L_{eq}$ , compared with the values estimated by the well-known simplified estimation method.

Figure 2.2.7 shows a comparison between the theoretically estimated curves using the method proposed in Section 2.2.2 and the experimentally sampled points in the cumulative distribution form of the noise level fluctuation for Case A. Moreover, Fig.2.2.8 shows the es-

-88-



Fig.2.2.6 A comparison between experimentally sampled points and theoretically estimated curves using Eq.(2.2.13) for the cumulative distribution form of the noise intensity fluctuation (Case B). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.2.13); ----, the second approximation (and the third approximation).



Fig.2.2.7 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case A). Experimentally sampled points are marked by • and theoretically estimated curves are respectively shown as ----, the first approximation using  $L_{eq}$  and  $\sigma_x^2$ ; ----, the second approximation using  $L_{eq}$ ,  $\sigma_x^2$  and  $L_{50}$ ; ----, the third approximation using  $L_{eq}$ ,  $\sigma_x^2$ ,  $L_{10}$  and  $L_{90}$ .

Measured val of Lea (dB)	ue Estimated value using Eq. (2.2.5) (dB)	Estimated values using the proposed method (dB)	L <sub>x</sub> levels used in the proposed method
75.0	74.8	74.8	Lio
		74.9	L 50
		75.0	L10, L50
		74.9	L50, L90

### Table 2.2.1 The estimated results of Leg using the proposed method (CaseA).

Table 2.2.2 The estimated results of Leg using the proposed method (CaseB).

Measured of Leg (dB)	value	Estimated value using Eq. (2.2.5) (dB)	Estimated values using the proposed method (dB)	L <sub>x</sub> levels used in the proposed method
73.7	-	75.5	74.6	Lio
			74.7	Lso
			74.4	L10, L50
			74.3	L5, L10

timated results for Case B. From these figures, it is obvious that the successive addition of higher expansion terms due to the reference levels of  $L_x$  moves the values estimated theoretically using the proposed method closer to the values measured experimentally.

To confirm the effectiveness of the proposed estimation method, road traffic noise data was also measured near an expressway in a rural district where the traffic volume was about 300 vehicles per hour. Table 2.2.3 shows the estimated results using the estimation method proposed in Section 2.2.1. Furthermore, in consideration of the effect of background noises, the estimated results for the original noise level distribution above  $L_{90}$  for Case B using the estimation method proposed in Section 2.2.2 are shown in Fig.2.2.9.

### 2.4 Conclusion

In this chapter, a general theory for the mutual relationship between several types of noise evaluation indices in relation to two typical indices,  $L_x$  and  $L_{eq}$ , has been proposed for the purpose of systematic evaluation of environmental noise. More specifically, a general method for estimation of  $L_{eq}$  using several specific  $L_x$  has been first proposed in a generalized form with inclusion of a well-known simplified estimation method derived under the assumption of an ordinary Gaussian distribution as a special case. Next, a new trial estimating the original noise level distribution using the measured values



Fig.2.2.8 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case B). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation using  $L_{eq}$  and  $\sigma_x^2$ ; ---, the second approximation using  $L_{eq}$ ,  $\sigma_x^2$  and  $L_{50}$ ; ---, the third approximation using  $L_{eq}$ ,  $\sigma_x^2$ ,  $L_{10}$  and  $L_{90}$ .

Measured	value	Estimated value	Estimated values	L <sub>x</sub> levels used
of Lea (dB)		using Eq. (2.2.5) (dB)	using the proposed method (dB)	in the proposed method
72.5	68.4	73.2	Lio	
		73.7	Lso	
		73.3	L5, L10	

Table 2.2.3 The estimated results of  $L_{eq}$  using the proposed method.



Fig. 2.2.9 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form above the level value of  $L_{90}$  (Case B) by considering the effect of background noises. Experimentally sampled points are marked by • and theoretically estimated curves are respectively shown as ----, the first approximation using  $L_{eq}$  and  $\sigma_x^2$ ; ---, the second approximation using  $L_{eq}$ ,  $\sigma_x^2$  and  $L_{50}$ ; ---, the third approximation using  $L_{eq}$ ,  $\sigma_x^2$ ,  $L_{10}$  and  $L_{50}$ .

of  $L_{eq}$  and several specific  $L_x$  has been proposed. The effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data. The following points remain for future studies. First, this method must be applied to many other actual cases to broaden and confirm its practical effectiveness. It is necessary that similar mutual relationships be found with other noise evaluation indices such as  $L_{NP}$ , TNI, WECPNL, etc. Chapter 3 Further Investigation of Method for Estimation of Noise Probability Distribution by Use of  $\rm L_{eq}$  and Several Specific  $\rm L_{X}$ 

#### 3.1 Introduction

In order to grasp quantitatively the mutual relationship between two noise evaluation indices,  $L_{eq}$  and  $L_x$ , somewhat systematical studies have already been considered in Chapter 2, and a method for estimation of  $L_{eq}$  was proposed using several specific  $L_x$ , with attention paid to the noise probability expression on a decibel scale. In addition, a new trial for estimation of the original noise level distribution using the measured values of  $L_{eq}$  and several specific  $L_x$  was also proposed with attention paid to the noise probability expression on an intensity scale (i.e., statistical Laguerre series expansion). This method would appear to be principally applicable to any kind of random phenomena. From the viewpoint of noise evaluation, however, such a probability expression on an intensity scale should be reconsidered at the first stage of study, especially in the form matched to the acoustic measurement on a decibel scale.

In this chapter, a further investigation in the statistical analysis for generally estimating the original noise level distribution on a decibel scale is proposed by first introducing a statistical orthonormal series expansion on an intensity scale taking the lognormal distribution as the first expansion term. The effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data. The results estimated using the proposed method agree very well with the experimental values compared with the results estimated using the previous method.

3.2 Theoretical consideration

As touched on in the introduction, let us first introduce a statistical orthonormal series expansion, for the purpose of emplying the noise probability expression on an intensity scale in the form matched to the acoustic measurement on a decibel scale, taking the lognormal distribution as the first expansion term. The above expression is shown as

$$P(x) = \frac{1}{\sqrt{2 \pi \sigma x}} exp\left\{-\frac{1}{2}\left(\frac{1 n x - \mu}{\sigma}\right)^{2}\right\}$$
$$\cdot \left[1 + \sum_{n=3}^{\infty} C_{n} \phi_{n}(x)\right], \qquad (2.3.1)$$

where two distribution parameters,  $\mu$  and  $\sigma^2$ , are given by the mean and variance of the noise intensity fluctuation x, as follows<sup>(35)</sup>:

$$\mu \triangleq 1 \quad n \quad \frac{\langle x \rangle^2}{\sqrt{\langle x^2 \rangle}}, \qquad \sigma^2 \triangleq 1 \quad n \quad \frac{\langle x \rangle^2}{\langle x^2 \rangle}, \qquad (2.3.2)$$

and the expansion coefficient  $C_n$  is given by the orthonormal polynomial,  $\phi_n(x),$  as follows :

$$C_{\mathbf{n}} \triangleq \langle \phi_{\mathbf{n}} (\mathbf{x}) \rangle . \tag{2.3.3}$$

The orthonormal polynomial  $\phi_n(x)$  is determined by Schmidt's orthogonalization procedure  ${}^{(36)}$ :

$$\phi_{n}(x) = \sum_{j=0}^{n} \lambda_{nj} x^{j},$$
 (2.3.4)

where the orthogonalization coefficient  $\lambda_{\mbox{nj}}$  is defined as

$$\lambda_{nj} = (d_{n} \cdot d_{n-1})^{-1/2}$$

$$\cdot ((n+1, j+1) \text{ cofactor of } d_{n}(x))$$
(2.3.5)

with

$$d_{n} \triangleq \begin{vmatrix} (x^{0}, x^{0}) & (x^{0}, x^{1}) \cdots (x^{0}, x^{n}) \\ (x^{1}, x^{0}) & (x^{1}, x^{1}) \cdots (x^{1}, x^{n}) \\ \vdots & \vdots & \vdots \\ (x^{n}, x^{0}) & (x^{n}, x^{1}) \cdots (x^{n}, x^{n}) \end{vmatrix}$$

$$d_{n}(x) \triangleq \begin{vmatrix} (x^{0}, x^{0}) & (x^{0}, x^{1}) \cdots (x^{0}, x^{j}) \cdots (x^{0}, x^{n}) \\ (x^{1}, x^{0}) & (x^{1}, x^{1}) \cdots (x^{1}, x^{j}) \cdots (x^{1}, x^{n}) \\ \vdots & \vdots & \vdots \\ (x^{n-1}, x^{0}) & (x^{n-1}, x^{1}) \cdot (x^{n-1}, x^{j}) \cdots (x^{n-1}, x^{n}) \\ x^{0} \cdots x^{1} \cdots x^{j} \cdots x^{n} \end{vmatrix}$$

(2.3.6)
Here, each element of the above matrices is given by

$$(x^{i}, x^{j}) \triangleq \int_{0}^{\infty} d(x) x^{i} x^{j} dx$$
  
=  $\int_{0}^{\infty} x^{i+j} \frac{1}{\sqrt{2\pi\sigma x}} e^{x} p\left\{-\frac{1}{2}\left(\frac{1 n x - \mu}{\sigma}\right)^{2}\right\} dx$   
=  $e^{x} p\left\{(i+j) \mu + \frac{1}{2}(i+j)^{2} \sigma^{2}\right\}$  (2.3.7)

with

d (x) 
$$\triangleq \frac{1}{\sqrt{2 \pi \sigma x}} e x p \left\{ -\frac{1}{2} \left( \frac{1 n x - \mu}{\sigma} \right)^2 \right\}$$
 (2.3.8)

Since  $L_{eq}$  is originally evaluated from the averaged noise intensity, two distribution parameters,  $\mu$  and  $\sigma^2$  can be rewritten using  $L_{eq}$  and  $\sigma_x{}^2$  (variance of x), as follows :

$$\mu = 2 \ln x_{0} + \frac{\ln 1 0}{5} L_{eq} - \frac{1}{2} \ln \left\{ \sigma_{x}^{2} + x_{0}^{2} \cdot 1 0^{\frac{Leq}{5}} \right\},$$
  
$$\sigma^{2} = \ln \left\{ \sigma_{x}^{2} + x_{0}^{2} \cdot 1 0^{\frac{Leq}{5}} \right\} - 2 \ln x_{0} - \frac{\ln 1 0}{5} Leq. \qquad (2.3.9)$$

The cumulative distribution function Q(x) connected directly with an arbitrary  $L_x$ , is formulated from Eq.(2.3.1) and its definition, as follows :  $1\pi x - \mu$ 

$$Q(x) \triangleq \int_{0}^{x} P(\xi) d\xi = \int_{-\infty}^{\sigma} N(\xi; 0, 1) d\xi + \sum_{n=3}^{\infty} \sum_{j=0}^{n} C_{n}\lambda_{nj} exp\{-2 j \mu - j^{2}\sigma^{2}\} + \sum_{n=3}^{\infty} \sum_{j=0}^{\sigma} C_{n}\lambda_{nj} exp\{-2 j \mu - j^{2}\sigma^{2}\} + \sum_{n=3}^{\sigma} \sum_{j=0}^{\sigma} N(\xi; 0, 1) d\xi$$
(2.3.10)

N 
$$(\xi; 0, 1) \triangleq \frac{1}{\sqrt{2\pi}} \exp\{-\xi^2/2\}$$
. (2.3.11)

From the definition of  $L_X$  and Eq.(2.3.10), the following relationship can be easily obtained :

$$1 - \frac{x}{1 \ 0 \ 0} = \int_{-\infty}^{\frac{1}{\pi} \frac{x}{x} - \mu} N \ (\xi \ ; \ 0 \ , \ 1 \ ) \ d \ \xi + \sum_{n=3}^{\infty} \sum_{J=0}^{n} C_{n} \lambda_{nJ} \ e \ x \ p \ \{-2 \ j \ \mu - j \ ^{2} \sigma \ ^{2}\} \frac{1 \ n x x - \mu - J \ \sigma^{2}}{\sigma} (2.3.12) \cdot \int_{-\infty}^{-\infty} N \ (\xi \ ; \ 0 \ , \ 1 \ ) \ d \ \xi \ ,$$

where  $x_X$  denotes the intensity value corresponding to  $L_X$ , as follows :

$$X_{x} = X_{0} \cdot 1_{0} L_{x/10}$$
(2.3.13)

Let  $x_{XR}$  (n=1,2,...,N) be the intensity values corresponding to the specific  $L_{XR}$  (n=1,2,...,N). After applying these  $x_{XR}$ (n=1,2,...,N) values to Eq.(2.3.12), the N-dimensional simultaneous equations can be constructed by regarding the expansion coefficient  $C_n$  (n=3,4,..., N+2) as unknown parameters. From the estimated values of  $\mu$ , $\sigma^2$  and  $C_n$ through the above procedures, one can estimate the original cumulative distribution function after substituting these values into Eq.(2.3. 10). The objective cumulative distribution form Q(L) on a decibel scale can of course be evaluated using the probability measure pre-

with

serving transformation :

Q (L) = Q (X)  
x = x 
$$_{0} \cdot 1 \ 0^{L/10}$$
. (2.3.14)

From the practical point of view, the explicit expressions for estimating  $\rm C_n$  for cases with N=1 and N=2 are shown as

(1) N=1 (reference level :  $L_{x1}$ ) :

$$C_{3} = \frac{G_{1} (X_{x1}; \mu, \sigma^{2})}{F_{1} (X_{x1}; \mu, \sigma^{2})} .$$
(2.3.15)

(2) N=2 (reference levels :  $\text{L}_{x1}$  and  $\text{L}_{x2})$  :

$$C_{3} = \frac{G_{1}(X_{x1}; \mu, \sigma^{2}) F_{2}(X_{x2}; \mu, \sigma^{2}) - G_{2}(X_{x2}; \mu, \sigma^{2}) F_{2}(X_{x1}; \mu, \sigma^{2})}{F_{1}(X_{x1}; \mu, \sigma^{2}) F_{2}(X_{x2}; \mu, \sigma^{2}) - F_{1}(X_{x2}; \mu, \sigma^{2}) F_{2}(X_{x1}; \mu, \sigma^{2})},$$

$$C_{4} = \frac{G_{2}(X_{x2}; \mu, \sigma^{2}) F_{1}(X_{x1}; \mu, \sigma^{2}) - G_{1}(X_{x1}; \mu, \sigma^{2}) F_{1}(X_{x2}; \mu, \sigma^{2})}{F_{1}(X_{x1}; \mu, \sigma^{2}) F_{2}(X_{x2}; \mu, \sigma^{2}) - F_{1}(X_{x2}; \mu, \sigma^{2}) F_{2}(X_{x1}; \mu, \sigma^{2})},$$

$$(2.3.16)$$

where  $F_1(\xi;\mu,\sigma^2)$ ,  $F_2(\xi;\mu,\sigma^2)$ ,  $G_1(\xi;\mu,\sigma^2)$  and  $G_2(\xi;\mu,\sigma^2)$  are respectively defined as

$$F_{1} (\xi ; \mu, \sigma^{2}) \triangleq \sum_{j=0}^{3} \lambda_{3j} e x p (-2 j \mu - j^{2} \sigma^{2})$$

$$\frac{\ln \ell - \mu - j \sigma^{2}}{\sigma}$$

$$\cdot \int_{-\infty}^{\sigma} N (\xi ; 0, 1) d\xi,$$

$$F_{2}(\xi ; \mu, \sigma^{2}) \triangleq \sum_{j=0}^{4} \lambda_{4j} \exp(-2j\mu - j^{2}\sigma^{2})$$

$$\cdot \int_{-\infty}^{\frac{\ln \xi - \mu - j\sigma^{2}}{\sigma}} N(\xi ; 0, 1) d\xi,$$

 $G_{1}(\xi; \mu, \sigma^{2}) \triangleq 1 - \frac{X_{1}}{100} - \int_{-\infty}^{\frac{\ln \xi - \mu}{\sigma}} N(\xi; 0, 1) d\xi,$ 

$$G_{2}(\xi; \mu, \sigma^{2}) \triangleq 1 - \frac{X_{2}}{100} - \int_{-\infty}^{\frac{\ln \xi - \mu}{\sigma}} N(\xi; 0, 1) d\xi.$$

(2.3.17)

## 3.3 Experimental consideration

In this section, the proposed method is applied to actual road traffic noise data to confirm its effectiveness. The arrangements for measuring road traffic noise at an observation point is shown in Fig.2.3.1. The road traffic noise data was recorded in advance using a data recorder in a measurement time interval of 2 hours. The data naturally exhibited non-stationary properties and has therefore been divided into two data groups with different fluctuation patterns of 1 hour, with sampling periods of 1 second. Let the first measurement interval be " Case A " and the second " Case B ".

Figures 2.3.2 and 2.3.3 show the estimated results for the



Fig.2.3.1 Arrangements for measuring road traffic noise at an observation point.

original noise level distribution for Case A. They show the comparisons between the experimentally sampled points and the theoretically estimated curves using both the proposed method and the previous one in the form of a cumulative distribution form on a decibel scale. From these estimated results, it can be seen that the results estimated using the proposed method agree well with the experimental values, as compared with the results estimated using the previous method. Moreover, the comparisons between the experimentally sampled points and the theoretically estimated curves for Case B are shown in Figs.2.3.4 and 2.3.5. Similarly, the results estimated using the proposed method agree well with the

## 3.4 Conclusion

In this chapter, a new trial for estimation of noise level distribution has been proposed using the measured values of  $L_{eq}$  and several specific  $L_x$ , especially in close connection with the practical situation of actual acoustical measurement on a decibel scale. More specifically, a statistical orthonormal series expansion taking a lognormal distribution as the first expansion term has been first introduced for the purpose of employing the noise probability expression on an intensity scale especially in relation to the actual acoustic measurement on a decibel scale. The effect of the statistical properties of  $L_{eq}$  and several specific  $L_x$  on the above noise



Fig.2.3.2 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case A).



Fig.2.3.3 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case A).



Fig.2.3.4 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case B).



Fig.2.3.5 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case B).

probability expression has been reflected in two distribution parameters,  $\mu$  and  $\sigma^2$ , and the expansion coefficient  $C_n$ . Finally, the effectiveness of the proposed method has been experimentally confirmed by applying it to two kinds of actual road traffic noise data through comparisons of estimation accuracy between the proposed method and the previous one.

This research is still in the early stages so the work reported here has only focussed on its methodological aspects. Accordingly, several future problems remain. This method must be applied to many other actual problems to broaden and confirm its practical effectiveness. Furthermore, the variance on an intensity scale used here should be estimated from the other evaluation indices such as  $L_x$  and  $L_{eq}$ , as it is not usually employed in actual noise evaluation and regulation problems. Chapter 4 Calculation of  $L_X$  and  $L_{eq}$  Noise Evaluation Indices by Use of Moment Statistics and Their Microcomputer-Aided On-Line Measurement

### 4.1 Introduction

As stated in the previous chapters, two noise evaluation indices,  $L_x$  and  $L_{eq}$ , play an important role in the field of noise evaluation and regulation problems. In order to evaluate these two indices, the usual measurement methods are given according to the original definition of each. That is, an  $L_x$  noise evaluation index is defined as the noise level exceeded by x percent throughout a total measurement time interval. In contrast, an  $L_{eq}$  noise evaluation index is defined as a constant noise level with a noise intensity value equal to the averaged intensity of the noise level fluctuation over a measurement time interval. The noise level fluctuation is generally measured these days in a quantized amplitude form at every discrete time period with a digital-type instrument. In cases in which this data is used to evaluate  $L_x$  and  $L_{eq}$  evaluation indices, the following fundamental problems still remain as seen from the viewpoint of signal processing :

(1) In order to evaluate  $L_{eq}$ , it is necessary that many level data with a fairly fine sampling period be obtained, since the noise intensity fluctuation after anti-logarithmic transformation of the sampled level datum fluctuates with large excursions, as compared with the original decibel-scaled level fluctuation.

(2) In principle, the mean value of the noise intensity fluctuation should be given by the sample mean operation based on the noise intensity data, especially with an equally quantized intensity amplitude. If noise intensity fluctuation after transformation of the measured noise level fluctuation with an equally quantized level amplitude into the intensity scale through the antilogarithmic transformation is used, a calculation error of the intensity mean will occur because of the above level quantization. (3) In order to evaluate  $L_x$ , a sampling period of an appropriate small The errors in  $L_X$  due to this sampling value must be selected. period have already been investigated by many researchers (e.g., see Refs.(37) and (38)). Furthermore, the value of  $L_X$  when x is small (e.g.,  $L_1$  and  $L_5$ ) is statistically unstable in comparison with that of the median, since the amount of data exceeding x percent is small. (39), (40)

In relation to the above problems, according to current measurement techniques, the sampling period may greatly influence the accuracy of the measurement result. In the case of the accuracy of Leq especially, a fine sampling period related to the time constant of the integration giving the noise level will generally give a good approximation of the results obtained with true integration. If a measurement system is constructed therefore with the use of a microcomputer according to current practice, a huge memory capacity is necessary for long-term measurement with such a fine sampling period. As mentioned in Chapter 2, explicit utilization of the statistical information on the decibel-scaled level fluctuation itself with the aid of the theory for estimating  $L_x$  and  $L_{eq}$  noise evaluation indices is very convenient (e.g., see Ref.(26)). The above statistical information is fairly stable and reliable, being based on the averaging operation supported by a large amount of data. In order to extract the statistical information, the sampling period of the noise level fluctuation can be greater than in current practice, refered to above. At the same time, each order moment statistics can be obtained successively by introducing an iterative calculation process. The problem of a huge memory capacity can therefore be solved by using this procedure.

In this chapter, a unified method for measuring two noise evaluation indices,  $L_x$  and  $L_{eq}$ , based on the theoretical study of Chapter 2 is proposed, using statistical information on the decibel-scaled level fluctuation under consideration. Based on this estimation method, a specific on-line measurement system for sequential measurement is constructed by use of a digital sound level meter and a microcomputer. Finally, the effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise.

4.2 Theoretical consideration

4.2.1 General algorithm for estimating Leq

Let us consider the noise level fluctuation L of an arbitrary non-Gaussian distribution type. According to the previous study in Chapter 2, a generalized expansion type expression for estimating  $L_{eq}$ is given by using the cumulant statistics of L, as follows :

$$L_{eq} (\triangleq 1 \ 0 \ 1 \ 0 \ g_{10} \langle x \rangle / x_0)$$

$$= \mu_{\rm L} + 0.115 \sigma_{\rm L}^{2} + 8.84 \times 10^{-3} \kappa_{\rm L3} + 5.09 \times 10^{-4} \kappa_{\rm L4} + \cdots$$

From Eq.(2.4.1), it is possible to generally estimate an  $L_{eq}$  noise evaluation index by reflecting not only lower order cumulants but also higher order cumulants in a hierarchical form.

To establish an evaluation algorithm with the aid of a microcomputer, the cumulant  $\kappa_{Ln}$ (n=1,2,...) must be obtained by spending a small amount of its memory. To achieve this, the mth order moment of L can first be calculated by means of the following iterative process :

$$\langle L^{\mathbf{m}} \rangle_{\mathbf{N}} = \frac{\mathbf{N} - 1}{\mathbf{N}} \langle L^{\mathbf{m}} \rangle_{\mathbf{N} - 1} + \frac{1}{\mathbf{N}} L_{\mathbf{N}}^{\mathbf{m}}.$$
 (2.4.2)

From Eq.(2.4.2), the mth order moment  $\langle L^m \rangle_N$  at the Nth measurement time based on the memorized past value of the mth order moment,  $\langle L^m \rangle_{N-1}$ , at the (N-1)th measurement time and the present datum,  $L_N$ , at the Nth measurement time can be obtained in an iterative form.

After the mth order moment within the specific measurement time interval using the above procedure has been obtained, the resultant moment,  $\langle L^n \rangle$  (n=1,2,...,m), can be transformed into the cumulant  $\kappa_{Ln}$ 

 $(n=1,2,\ldots,m)$ , as follows <sup>(41)</sup>:

$$\kappa_{L1} = \langle L \rangle , \qquad \kappa_{L2} = \langle L^2 \rangle - \langle L \rangle \kappa_{L1},$$

$$\kappa_{L3} = \langle L^3 \rangle - 2 \langle L \rangle \kappa_{L2} - \langle L^2 \rangle \kappa_{L1},$$

$$\kappa_{L4} = \langle L^4 \rangle - 3 \langle L \rangle \kappa_{L3} - 3 \langle L^2 \rangle \kappa_{L2} - \langle L^3 \rangle \kappa_{L1},$$

$$\kappa_{L5} = \langle L^5 \rangle - 4 \langle L \rangle \kappa_{L4} - 6 \langle L^2 \rangle \kappa_{L3} - 4 \langle L^3 \rangle \kappa_{L2} \qquad (2.4.3)$$

$$- \langle L^4 \rangle \kappa_{L1}, \qquad \cdots$$

The objective  $L_{eq}$  can thus be estimated by substituting the calculated value of cumulant statistics into Eq.(2.4.1).

4.2.2 General algorithm for estimating  $L_x$ 

In order to estimate the arbitrary  $L_x$  noise evaluation indices (e.g.,  $L_5$ ,  $L_{10}$ ,  $L_{50}$ ,  $L_{90}$ , ...), let us introduce a statistical Hermite series expansion as the cumulative distribution function universally applicable to the noise level fluctuation L of an arbitary non-Gaussian distribution type, as follows<sup>(13)</sup>:

$$Q (L) = \int_{-\infty}^{L} N (\xi ; \mu_{L}, \sigma_{L}^{2}) d\xi - \frac{\kappa_{L3}}{6 \sigma_{L}^{2}} N (L ; \mu_{L}, \sigma_{L}^{2}) \cdot H_{2} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) - \frac{\kappa_{L4}}{2 4 \sigma_{L}^{3}} N (L ; \mu_{L}, \sigma_{L}^{2}) H_{3} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) - \frac{\kappa_{L5}}{1 2 0 \sigma_{L}^{4}} N (L ; \mu_{L}, \sigma_{L}^{2}) H_{4} \left( \frac{L - \mu_{L}}{\sigma_{L}} \right) - \cdots$$
(2.4.4)

The value of the nth order Hermite polynomial,  ${\rm H}_n(\,.\,)\,,$  can be

calculated by the following recursive formula suitable for computer processing  $\binom{42}{2}$ :

$$H_{n}(\xi) = \xi H_{n-1}(\xi) - (n-1) H_{n-2}(\xi).$$
(2.4.5)

In addition, the calculation of the first expansion term of Eq.(2.4.4) can easily be performed by using the well-known error function, as follows :

$$\int_{-\infty}^{L} N \ (\xi \ ; \ \mu_{\rm L}, \ \sigma_{\rm L}^{2}) \ d\xi = 1 - \frac{1}{\sqrt{\pi}} \ {\rm E} \ {\rm r} \ {\rm f} \ c \left(\frac{L - \mu_{\rm L}}{\sqrt{2} \sigma_{\rm L}}\right) \ , \qquad (2.4.6)$$

where Erfc(.) denotes the error function defined as

$$\operatorname{Erfc}(\xi) \triangleq \int_{e}^{\infty} e x p (-t^{2}) d t . \qquad (2.4.7)$$

At this time, the value of Erfc(.) can be accurately calculated by the following approximation<sup>(43)</sup>:

$$\frac{2}{\sqrt{\pi}} \operatorname{Erfc}(\xi) = \frac{1}{[1+0.0705230784\xi + 0.0422820123\xi^{2}]} + 0.0092705272\xi^{3} + 0.0001520143\xi^{4}] + 0.0002765672\xi^{5} + 0.0000430638\xi^{6}]^{16}.$$
(2.4.8)

Since the relationship between  $\mathsf{Q}(\mathsf{L})$  and  $\ \mathsf{L}_{\mathsf{X}}$  is given by

$$1 - \frac{X}{1 \ 0 \ 0} = Q \ (L_{x}) \tag{2.4.9}$$

and Q(L) is a monotone-increasing function, the objective  $L_X$  can be directly evaluated from the estimated curve of Q(L) using the linear interpolation, even for measurement with a fairly large discrete level amplitude. This procedure is quite convenient for economizing calculation time, compared to the usual numerical integration. Consequently, after statistical information on the noise level fluctuation has been obtained in the manner described in the previous section, an arbitrary  $L_X$  can be calculated by this method. Furthermore, based on this estimation method, all the data of the noise level fluctuation need not be memorized and problem can be solved with a small amount of memory capacity when using a microcomputer for the measurement system.

4.3 Experimental consideration

4.3.1 Confirmation of effectiveness of the proposed estimation method

For the purpose of confirming the effectiveness of the proposed estimation method, a measurement system has been constructed using a digital sound level meter and a portable microcomputer. The block diagram of this measurement system is shown in Fig.2.4.1.

Two kinds of road traffic noise have been measured as typical examples of environmental noise. One was measured near a national main road in a large city with a traffic volume of 244 vehicles per 10 minutes, and the other near a country road in a rural district with a



Fig.2.4.1 Block diagram of measurement system with the aid of a portable microcomputer.

traffic volume of 33 vehicles per 10 minutes. The measurement time interval and its sampling period were selected to be 10 minutes and 1 second, respectively. Here, let us define the former case as "Case A " and the latter as "Case B ". Figure 2.4.2 shows the arrangements for measuring road traffic noise at an observation point for Case A, and Fig.2.4.3 the arrangements for Case B.

Table 2.4.1 shows the estimated results of Leg for Case A using the proposed estimation method. Here, let us define the expansion expression of Eq.(2.4.1) from the first expansion term corresponding to the well-known simplified estimation method to the term due to the nth order cumulant, as the (n-1)th approximation of the estimated  $L_{eq}$ . In addition, the measured value of  $L_{eq}$  in this table shows the value measured by a precision integrating sound level meter for Leg. According to these results, the calculated value, using a well-known simplified estimation method derived under assumption of a well-known Gaussian distribution, agrees approximately with the measured value. More specifically, however, the estimated values using the proposed method tend to agree well with the successive addition of higher order expansion terms. Table 2.4.2 shows the estimated results for Case B. According to these results, the estimation accuracy using the well-known simplified estimation method is not sufficient for evaluation. It is clear, though, that the successive addition of higher order expansion terms moves the values estimated theoretically using the proposed method closer



Fig.2.4.2 Arrangements for measuring road traffic noise at an observation point (Case A).



Fig.2.4.3 Arrangements for measuring road traffic noise at an observation point (Case B).

Measured value	Estima	ies using the		
of Leg (dB)	proposed method (dB)			
85.9	86.4(the	first	approximation)	
	86.0(the	second	approximation)	
	86.1(the	third	approximation)	
	86.1(the	fourth	approximation)	

# Table 2.4.1 The estimated results of Leg using the proposed method (CaseA).

Table 2.4.2 The estimated results of  $L_{eq}$  using the proposed method (CaseB).

Measured value of Leg (dB)	Estimated values using the proposed method (dB)				
80.7	77.7(the 82.0(the 82.8(the	first second third	approximation) approximation) approximation)		
	81.0 (the	fourth	approximation)		

to the values measured experimentally.

Figure 2.4.4 shows a comparison between the theoretically estimated curves using the proposed method and the experimentally sampled points in the cumulative distribution form of the noise level fluctuation for Case A (arbitrary  $L_X$  noise levels are directly extracted by this distribution form). From this figure, the theoretically estimated curve due to the first approximation (i.e., Gaussian distribution) agrees approximately with the experimentally sampled points. More precisely, however, the estimated values using the proposed method tend to agree well with the measured values, with the successive addition of higher order expansion terms. Figure 2.4.5 shows the estimated results for Case B. In this figure, it can be seen that curve theoretically estimated by using only the first expansion term corresponding to the Gaussian distribution does not agree well with the experimentally sampled points. It is clear that the successive addition of higher order expansion terms moves the values estimated theoretically using the proposed method closer to the values measured experimentally. In each case, the estimation accuracy for both  $L_x$  and  $L_{eq}$  noise evaluation indices using the proposed estimation method is clearly sufficient.

4.3.2 Construction of an on-line measurement system for sequential measurement

In this section, an on-line measurement system for the sequential



Fig.2.4.4 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case A). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.4.4); ----, the second approximation (and the third and fourth approximations).



Fig.2.4.5 A comparison between experimentally sampled points and theoretically estimated curves using the proposed method for the cumulative distribution form of the noise level fluctuation (Case B). Experimentally sampled points are marked by  $\bullet$  and theoretically estimated curves are respectively shown as ----, the first approximation of Eq.(2.4.4); ----, the second approximation (and the third approximation); -----, the fourth approximation.

measurement of two noise evaluation indices,  $L_X$  and  $L_{eq}$ , based on the proposed estimation method is constructed. According to the estimated results shown above, in the actual estimation, the estimated values due to the fourth approximation have been employ-Since calculation time is needed for estimation of the ed. two indices, concurrent processing by two microcomputers can be A block diagram of this measurement system is shown in used. Fig.2.4.6. Microcomputer A is used for the iterative process for extracting the moment statistics within each measurement time interval using Eq.(2.4.2) and the control system for the communication lines between the digital sound level meter and Microcomputer B with an RS-232C type interface. Microcomputer B is used for calculation of  $L_x$  and  $L_{eq}$  using the proposed estimation method after transforming the moment statistics  $\langle L^n \rangle$  into the cumulant statistics  $\kappa_{L,n}$ . Each measurement time interval can be arbitrarily selected using the clock signal generator in Microcomputer A. In this measurement, each measurement time interval has been selected at 10 minutes. Two noise evaluation indices,  $L_x$  and  $L_{eq}$ , for the road traffic noise (Case A) have been measured for 24 hours. Figure 2.4.7 shows the actual location at which road traffic noise was measured with the proposed on-line measurement system. Figure 2.4.8 shows the estimated results of  $L_5$ ,  $L_{10}$ , L50, L90 and Leg using the proposed measurement system. The proposed system performed well for long-term measurement in each measurement time interval.



Fig.2.4.6 Block diagram of on-line measurement system for  $\rm L_{\rm X}$  and  $\rm L_{eq}$  using the proposed method.



Fig.2.4.7 Actual scene of road traffic noise measurements using the proposed on-line measurement system.



Fig.2.4.8 The estimated results of  $\rm L_X$  and  $\rm L_{eq}$  using the proposed on-line measurement system.

## 4.4 Conclusion

Based on the previous study in Chapter 2, a unified method for measuring two noise evaluation indices, Lx and Leg has been proposed in this chapter using statistical information on noise level fluctuation. The proposed method is generally applicable to any kind of random phenomena of a non-Gaussian property. It has been derived in a generalized form including a well-known simplified estimation method derived under the assumption of a well-known Gaussian dis-The statistical information used in the proposed method tribution. could be extracted successively by introducing an iterative calculation process, making it possible to economize microcomputer memory. In the proposed method, only a small amount of memory is required even when changing the measurement time interval. By using this unified theory, an on-line measurement system for sequential measurement through a long-term measurement time interval has been constructed using a digital sound level meter and a portable microcomputer.

Of course, this study is still in its early stage and has focussed only on its fundamental aspects. Problems remain in its application to other situations. CONCLUSION

1

•

### CONCLUSION

The main results of this doctoral thesis are here summarized.

As titled in this study, a statistical prediction method for the noise level distribution and the mutual relationship between various kinds of noise evaluation indices in close relation to  $L_X$  and  $L_{eq}$  have been discussed.

In Part I, a general method for prediction of the probability distribution form of environmental random noises has been considered.

In Chapter 1, a statistical method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow has been proposed. Previously, various approaches for predicting road traffic noise have been proposed from various points of view. These prediction methods can essentially be divided into the following two groups. One group comprises structural methods based on physical models. The other comprises functional methods based on mathematical models. Accordingly, in the previous study, some kind of hybrid method combining the advantages latent in both prediction groups was proposed. In order to establish this kind of prediction method, a new approach equivalent to that used for an idealized Poisson type traffic flow has been proposed in this chapter for practical use. The proposed prediction method can treat any kind of road traffic noise fluctuation generated from actual non-Poisson type traffic flow, once information has been obtained on the elementary response wave form of the level time pattern observed when one vehicle passes directly in

front of an observation point. The effectiveness of the proposed prediction method has been experimentally confirmed by applying it to actually observed road traffic noise data.

In Chapter 2, a practical method for prediction of road traffic noise at a T-type road intersection has been proposed based on the image method. In Chapter 1, a general method for prediction of road traffic noise generated from an arbitrary non-Poisson type traffic flow was proposed based on the information on the elementary response wave form observed experimentally. In order to establish a more effective prediction method, a theoretical method for estimating the above elementary response wave form must be found, especially in a complicated realistic acoustical environment. In this chapter, a new approach for estimating theoretically this elementary response wave form has been proposed for the case of a T-type road intersection with sound insulation barriers, using a well-known image method. The effectiveness of the proposed method has been experimentally confirmed by applying it to the road traffic noise data observed at a T-type road intersection in a city area.

In Chapter 3, a study on the stochastic response of a sound insulation barrier has been discussed. In the practical engineering field of noise control, a sound insulation barrier is very often constructed to produce attenuation of environmental random noises such as the road traffic noise discussed in the previous chapters. The acoustical design and/or the evaluation problem of barriers have already been considered by many researchers. Almost all these studies, however, were confined only to the effects on deterministic signals or the gross average evaluation of shielding effects. In this chapter, a prediction method of the stochastic insulation effect of a sound insulation barrier has been proposed for the case of an arbitrary probability distribution for the random noise incident on the barrier. The emphasis in this chapter has been focussed on how the output noise distribution form using information on the statistical properties of the random input noise and the frequency characteristics of the sound insulation barrier can be predicted. The effectiveness of the proposed prediction method has been experimentally confirmed using actual noise data.

In Part II, in order to evaluate precisely complicated real environmental random noise, the mutual relationship between various kinds of statistical evaluation indices in relation to  $L_x$  and  $L_{eq}$  has been considered. In the field of noise evaluation and regulation problems, two noise evaluation indices,  $L_x$  and  $L_{eq}$ , play an important role. Accordingly, the establishment of a general theory for the above mutual relationship is of basic importance to obtain systematical evaluation of actual environmental random noises. Previously, various studies for finding mutual relationships of this kind have already been considered by many researchers. Almost all of these studies, however, were confined to approximate methods derived under the assumption of an ordinary Gaussian distribution and/or practical methods derived by applying the conventional linear regression analysis method to actually observed data. From the above practical points of view, a general theory for mutual relationships between various noise evaluation indices in close relation to  $L_x$  and  $L_{eq}$  has been presented in Part II.

In Chapter 1, a method for estimation of an  $L_{eq}$  noise evaluation index has been proposed by use of moment statistics of the noise level fluctuation of arbitrary non-Gaussian distribution type. To estimate the value of  $L_{eq}$  for an arbitrary non-Gaussian type random noise fluctuation, a new approach equivalent to that used for the Gaussian type has been first proposed for practical application. The proposed estimation method was given in a generalized form universally applicable to any kind of random phenomena of a non-Gaussian property, including a well-known simplified expression derived under assumption of an ordinary Gaussian distribution. Finally, it has been applied to actual noise data to confirm its effectiveness.

In Chapter 2, the mutual relationship between several statistical indices connected with  $L_x$  and  $L_{eq}$  has been described. An estimation method of  $L_{eq}$  is presented using the known values of several specific  $L_x$  with the use of the mean and variance of the random noise fluctuation. The proposed estimation formula was also given in a generalized form including the well-known simplified expression mentioned above. Next, a new trial estimating the original distribution form of the random noise fluctuation has been presented using the known values of  $L_{eq}$  and several specific  $L_x$ . The effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data.

-128-

In Chapter 3, a method for estimation of the original distribution form of the random noise fluctuation has been also proposed. In this case, a more suitable expression of noise probability distribution on an intensity scale has been newly introduced in a form matched to acoustic measurement on a decibel scale. The effectiveness of the proposed method has been experimentally confirmed by applying it to actual road traffic noise data, and through comparisons of estimation accuracy between the proposed method and the previous one derived in Chapter 2.

In Chapter 4, based on the previous discussion in Chapter 2, a precise calculation method of  $L_x$  and  $L_{eq}$  noise evaluation indices has been proposed by use of information on the noise level fluctuation (i.e., moment statistics or cumulant statistics). Based on this estimation method, a specific on-line measurement system has been constructed with the aid of a microcomputer by introducing an iterative process for extracting the moment statistics of random noise fluctuation on a decibel scale. The effectiveness of the proposed method has been experimentally confirmed by applying it to actual measurement of road traffic noise.

This study is still in its early stages and the work presented here has focussed primarily on fundamental aspects. Many problems remain for future research. The following studies, for example, are left for the future :

(1) A reasonable evaluation method for complex situations in which environmental noise consists of various sound sources such as the road traffic noise, the aircraft noise, the construction noise, etc. is required.

- (2) A systematical evaluation method for various types of noise reduction systems in a room, such as sound absorbing materials, sound insulation walls, etc. must be considered.
- (3) It is also necessary that a reasonable method for evaluating more complicated real environmental noise attenuated by various reduction systems acoustically coupled each other be established.
- (4) More general probability expressions must be introduced by considering more severe actual phenomena in a form matched to acoustic measurements : i.e., non-stationary random processes, restricted fluctuation range, etc.
- (5) Based on the discussion described in this study, practical methods must be established for concrete countermeasures to noise pollution using actual noise reduction systems (sound absorbing materials, sound insulation walls, signal control systems for traffic flow, etc.).

#### ACKNOWLEDGMENTS

The author would like to express his deepest gratitude to his supervisor Professor Mitsuo Ohta for his valuable guidance and advice throughout the duration of this investigation. The author would also like to thank Associate Professor Kazutatsu Hatakeyama and Research Associate Dr. Akira Ikuta for their many valuable comments and suggestions.

The author would like to express his cordial thanks to Professor Eihachiro Nakamae, Professor Yho Yonezawa, Professor Koichi Mizukami, Professor Hiroshi Sasaki and Associate Professor Koji Ito for their helpful suggestions to the development of this thesis.

Thanks are also due to Professer Shizuma Yamaguchi of the Maritime Safety Academy, Associate Professor Hirofumi Iwashige of Hiroshima University, Professor Tadao Mine of Fukuyama University and Associate Professor Hikaru Shimizu of Fukuyama University for their kind advice. The author is greatful to Dr. Eiji Uchino of Kyushu Institute of Technology, Associate Professor Kiminobu Nishimura of Ube Technical College, Mr. Noboru Nakasako of Hiroshima Institute of Technology and Dr. Maomitsu Takaki of Japan Steel Works for their valuable suggestions.

The author appreciates the help of his colleagues, past and present, the Electronic Control Engineering Laboratory of Hiroshima University and the Electrical Fundamentals Engineering Laboratory of Fukuyama University. In particular, the author would like to thank Messrs. Kenzo Mikamoto, Fujinori Hirahara, Hakaru Hosoya, Yasuhiro Watanabe, Sumio Koresawa, Tomoyuki Katayama, Satoshi Shioji, Hiroshi Shirakawa and Yuzhou Yei for their helpful co-operation.

Finally, the author acknowledges President Shigeru Miyachi, Vice-President Kazuto Takayama and the teaching staff of Fukuyama University for their authorization of the author's transfer to the Systems Engineering Course of the Graduate School of Engineering of Hiroshima University.
## REFERENCES

- (1) M. A. Bruges, "Noise prediction for urban traffic condition related to measurements in the Sydney metropolitan area, "Applied Acoustics 10, pp.1-7 (1977).
- (2) M. E. Delaney, D.G. Harland, R. A. Hood and W. E. Scholes, "The prediction of noise levels  $L_{10}$  due to road traffic, "Journal of Sound and Vibration 48, pp.305-325 (1979).
- (3) K. Ishii, "Prediction of road traffic noise (part I) Method of practical calculation, "Journal of the Acoustical Society of Japan 31, pp.507-517 (1975) (in Japanese).
- (4) D. Revill, "Traffic noise measurement and prediction techniques," Noise Control, Vibration and Insulation Jan., pp.26-32 (1976).
- (5) K. Takagi, K. Hiramatsu, T. Yamamoto and K. Hashimoto, " Investigation on road traffic noise based on an exponentially distributed vehicles model, " Journal of Sound and Vibration 36, pp.417-431 (1974).
- (6) T. M. Barry and J. A. Reagan, "FHWA highway traffic noise prediction model, "FHWA-RD-77-108, U.S. Federal Highway Administration, Washington, D.C., B-1-B-5 (1978).
- (7) E. J. Rickley, D. W. Ford and R. W. Quinn, "Highway noise measurements for verification of prediction models," DOT-TSC-FHWA-78-1, U.S. Department of Transportation, Washington, D.C., A-1-D-174 (1978).

- (8) M. Uzu, Y. Sasaki, M. Nobuso and M. Ohta, " A practical hybrid method for prediction of road traffic noise from a viewpoint of filtered Poisson process model (in Japanese).
- (9) U. J. Kurze, "Statistics of road traffic noise," Journal of Sound and Vibration 18, pp.171-195 (1971).
- (10) M. Ohta, K. Hatakeyama, H. Iwashige and N. Yoshino, "Statistical theory on the general random noise of arbitrary distribution type with quantized level and its actual application, "Acustica 46, 201-209 (1980).
- (11) W. Feller, An Introduction to Probability Theory and Its Application (John Wiley and Sons, New York, 1957).
- (12) M. Ohta, S. Yamaguchi, S. Hiromistu and K. Hatakeyama, " A prediction theory for the level distribution of urban noise produced by the establishment of a new road and its experimental simulation, " Journal of the Acoustical Society of Japan 31, pp.72-80 (1975) (in Japanese).
- (13) M. Ohta and T. Koizumi, "General statistical treatment of the response of a non-linear rectifying device to a stationary random input, "Institute of Electrical and Electronic Engineers Transaction of Information Theory IT-14, pp.595-598 (1968).
- (14) S. Yamaguchi, M. Ohta and K. Nakamura, " A practical probabilistic prediction of road traffic noise from filtered Poisson process model with a simplified elementary time pattern of triangular type, " Archives of Acoustics 10, pp.75-91 (1985).
- (15) H. Kuttruff, " Zur Berechnung von Pegelmittelwerten und Schwan-

kungsgrößen bei Straßenlärm, " Acustica 32, pp.57-69 (1975).

- (16) M. Ohta and Y. Mitani, " A fundamental study for predicting the urban street noise by use of the image method approach, " Archives of Acoustics 10, pp.59-74 (1985).
- (17) Z. Maekawa, " Noise reduction by screens, " Applied Acoustics 1, pp.157-173 (1968).
- (18) M. Koyasu and M. Yamashita, "Scale model experiments on noise reduction by acoustic barrier of a straight line source," Applied Acoustics 6, pp.232-242 (1973).
- (19) C. W. Menge, "Highway noise : Sloped barriers as an alternative to absorptive barriers, "Noise Control Engineering 14, pp.74-78 (1980).
- (20) U. J. Kurze, "Noise reduction by barriers, " Journal of the Acoustical Society of America 55, pp.504-518 (1974).
- (21) K. Fujiwara, Y. Ando and Z. Maekawa, "Noise control by barriers (part 2) Noise reduction by an absorptive barrier, "Applied Acoustics 10, pp.167-179 (1977).
- (22) T. Isei, T. F. W. Embleton and J. E. Piercy, "Noise reduction by barriers on finite impedance ground, "Journal of the Acoustical Society of America 67, pp.46-58 (1980).
- (23) U. J. Kurze and G. S. Anderdon, "Sound attenuation by barriers, " Applied Acoustics 4, pp.35-53 (1971)
- (24) B. D. Lim and B. H. Lee, "Effect of absorptive surfaces on the sound diffraction by a right angle wedge, "Journal of the Acoustical Society of Japan (E) 8, pp.159-165 (1987).

- (25) C. M. Harris, Handbook of Noise Control (McGraw-Hill, New York, 1979).
- (26) C. G. Don and I. G. Rees, "Road traffic sound level distributions, "Journal of Sound and Vibration 100, pp.41-53 (1985).
- (27) A. Alexandre, J. P. Barde, C. Lamura and F. J. Langdon, Road Traffic Noise (Applied Science Publishers, London, 1975).
- (28) U. S. Environmental Protection Agency, Information on levels of environmental noise requisite to protect public health and welfare with an adequate margin of safety, 550/9-74-004 (1974).
- (29) R. W. B. Stephens and H. G. Leventhall, Acoustics and Vibration Progress : Volume 1 (Chapman and Hall, London, 1974).
- (30) J. E. Wesler, "Community noise survey of Medford Massachusetts," Journal of the Acoustical Society of America 54, pp.985-995 (1973).
- (31) S. Takeshita, H. Taguchi and T. Yasutomi, "Conversion of  $L_x$  data into  $L_{eq}$  on the basis of Weibull distribution, "Proceedings of the 1989 Meeting of the Institute of Noise Control Engineering of Japan, pp.317-320 (in Japanese).
- (32) G. W. Cermak and P. C. Cornillon, "Multidimensional analyses of judgments about traffic noise, "Journal of the Acoustical Society of America 54, pp.1412-1420 (1973).
- (33) M. Mori and H. Tsukaguchi, "Estimation of environmental noise by Weibull distribution, "Proceedings of the 1983 Meeting of the Institute of Noise Control Engineering of Japan, pp.77-80 (in Japanese).
- (34) K. Attenbrough, S. Clark and W. A. Utley, " Background noise

levels in the United Kingdom, "Journal of Sound and Vibration 48, pp.359-375 (1976).

- (35) H. J. Larson and B. O. Shubert, Probabilistic Models in Engineering Science, Vol. I, Random Variables and Stochastic Processes (John Wiley & Sons, New York, 1979).
- (36) H. Ogura, Theory of Random Processes (Corona, Tokyo, 1978) (in Japanese).
- (37) D. J. Fish, "Statistical sampling in community noise measurement," Journal of Sound and Vibration 30, 221-236 (1973).
- (38) J. G. Vaskor, S. M. Dickinson and J. S. Bradley, "Effect of sampling on the statistical descriptors of traffic noise," Applied Acoustics 12, pp.11-124 (1979).
- (39) R. E. Devor, P. D. Schomer, W. A. Kline and R. D. Neathamer,
  " Development of temporal sampling strategies for monitoring noise, " Journal of the Acoustical Society of America 66, pp.763-771 (1979).
- (40) R. J. Peppin and C. W. Rodman, Community Noise (American Society for Testing and Materials, Philadelphia, 1978).
- (41) E. Lloyd, Handbook of Applicable Mathematics, Volume II : Probability (John Wiley, New York, 1980).
- (42) S. Moriguchi, S. Utagawa and S. Hitotsumatsu, Mathematical Formula III (Iwanami, Tokyo, 1977) (in Japanese).
- (43) C. Hastings, Approximation for Digital Computers (Princeton University Press, New Jersey, 1955).

# LIST OF THE AUTHOR'S PUBLICATIONS

- M. Ohta, K. Hatakeyama and Y. Mitani, "A new method of estimating representative statistics of random noise by on-line treatment," Acustica, Vol.51, No. 6, pp.296-299 (1982); (Research Note).
- M. Ohta, S. Yamaguchi and Y. Mitani, "A new on-line estimation method of representative statistics of environmental random noise and vibration based on actual digital measurement," Journal of the Acoustical Society of Japan Vol. 39, No.6, pp. 361-366 (1983) (in Japanese).
- 3. M. Ohta, S. Yamaguchi and Y. Mitani, "A simplified probabilistic evaluation method for the sound insulation effect of barriers," Bulletin of Australian Acoustical Society Vol. 11, No. 3, pp. 97-103 (1983).
- 4. S. Yamaguchi, M. Ohta, K. Nishimura and Y. Mitani, "A general theory of the state probability for an arbitrary type of nonlinear stochastic energy system based on the additive property of energy and its application to noise environmental system," Proceedings of 1st IASTED International Symposium on Applied Control and Identification Vol.1, pp.9-5- 9-10 (1983, Copenhagen).
- 5. Y. Mitani and M. Ohta, "Probabilistic evaluation method of living environmental noise control system by use of modified statistical energy analysis method," Proceedings of 1983 International Conference on Noise Control Engineering Vol. 2, pp. 1179-1182 (1983, Edinburgh).
- M. Ohta, S. Yamaguchi and Y. Mitani, "A precise noise probability estimation from roughly observed data with quantized levels," Journal of the Acoustical Society of America Vol. 76, No. 1, pp. 122-127 (1984).
- 7. H. Iwashige, M. Ohta, and Y. Mitani, "A probabilistic evaluation of sound energy fluctuations in a room affected by the constant of instrumentation and various absorption materials," Proceedings of 1984 International Conference on Noise Control Engineering Vol. 2, pp. 1209-1212 (1984, Hawaii).
- 8. M. Ohta, Y. Mitani and T. Sumimoto, "A generalized theory for the mutual relationship among several type noise evaluation indices connected with  $L_{eq}$  and  $L_x$  and its experiment," Journal of the Acoustical Society of Japan Vol.

41, No. 9, pp. 598-607 (1985) (in Japanese).

- 8. Y. Mitani and M. Ohta, "Mutual relation among environmental noise evaluation indices on the basis of L<sub>x</sub> and L<sub>eq</sub> evaluation indices and its application to the actual road traffic noise," Proceedings of 12th International Congress on Acoustics Vol. 1, C4-4 (1986, Toronto).
- 9. M. Ohta and Y. Mitani, "A fundamental study for predicting the urban street noise by use of the image method approach," Archives of Acoustics Vol.10, No. 1, pp. 59-74 (1985).
- 9. M. Ohta, K. Hatakeyama and Y. Mitani, " A prediction of the street noise level over the typical city area by use of image method and simulation approaches," Proceedings of 1st Acoustics Conference of West Pacific Region, 2A-03-1 - 2A-03-13 (1982, Singapore).
- 10. M. Ohta and Y. Mitani, " A study on the probabilistic response of typical sound insulation systems inside and outside room with arbitrary random excitation, " Journal of the Acoustical Society of Japan (E) Vol. 7, No. 5, pp. 291-295 (1986) ; (Technical Report).
- M. Ohta and Y. Mitani, "A practical evaluation method for the stochastic noise reduction effect of sound absorbing materials," Acoustics Australia Vol. 14, No. 3, pp. 75-78 (1986).
- 12. M. Ohta and Y. Mitani, "The effect of the restricted fluctuation range in the actual random noise on the transformation of evaluation indices from L<sub>eg</sub> to L<sub>x</sub>." Journal of the Acoustical Society of Japan Vol.43, No.3, pp.143-151 (1987) (in Japanese).
- 13. M. Ohta and Y. Mitani, "A prediction method for road traffic noise generated from arbitrary non-Poisson type traffic flow based on an approach equivalent to that for a standard Poisson type traffic flow," Journal of Sound and Vibration Vol. 118, No. 1, pp. 11-21 (1978).
- 14. Y. Mitani and M. Ohta, "A new estimation method of noise evaluation index, L<sub>eq</sub>, by use of the statistical information on the arbitrary type noise level fluctuation," Journal of the Acoustical Society of Japan (E) Vol.9, No.1, pp. 47-51 (1988); (Technical Report).

- 14. Y. Mitani and M. Ohta, "A new trial of estimating an L eq evaluation index by use of the statistical information on the noise level fluctuation," Proceedings of 1986 International Conference on Noise Control Engineering Vol. 2, pp. 1275-1280 (1986, Cambridge).
- 15. Y. Mitani and M. Ohta, " A practical probabilistic evaluation method for the noise reduction effect of sound absorbing materials," Journal of the Acoustical Society of Japan (E) Vol.9, No.3, pp.147-152 (1988) ; (Technical Report).
- 16. M. Ohta and Y. Mitani, " A practical prediction method using an elementary response wave form for the road traffic noise at a T-type intersection," Journal of the Acoustical Society of Japan (E) Vol.9, No.4, pp.195-200 (1988); (Technical Report).
- 16. Y. Mitani and M. Ohta, "A practical prediction method using an elementary response wave form for the road traffic noise," Proceedings of 1987 International Conference on Noise Control Engineering Vol.1 (1987, Beijing).
- 17. Y. Mitani and M. Ohta, " A calculation of L and L eq noise evaluation indices by use of statistical information on the noise level fluctuation, and its microcomputer-aided on-line measurement," Applied Acoustics Vol. 25, pp. 33-47 (1988).
- 18. M. Ohta and Y. Mitani, "A practical prediction method for road traffic noise at a T-type road intersection based on the image method," Acustica Vol. 67, No. 2, pp. 152-156 (1988); (Research Note).
- 19. M. Ohta, A. Ikuta and Y. Mitani, "A stochastic signal processing of analog and digital types for acoustic random power state variable and an establishment of digital filter, "Transactions of the IEICE Vol.E72, No.5, pp.521-530 (1989).
- 20. M. Ohta and Y. Mitani, "Time series prediction method for stochastic acoustic signals by use of an adaptive function model," Measurement Vol.7, No. 2, pp. 68-76 (1989).
- 20. Y. Mitani and M. Ohta, "Time series modeling for acoustic signals by a generalized adaptive function and its application to actual random noise data,"

Proceedings of 1986 International Conference on Acoustics, Speech, and Signal Processing Vol. 4, pp. 2355-2358 (1986, Tokyo).

- 21. M. Ohta and Y. Mitani, "A statistical analysis for estimating the noise level probability distribution from L<sub>eq</sub> and L<sub>x</sub> noise levels, " Journal of the Acoustical Society of Japan (E) Vol.10, No.3, pp.175-180 (1989); (Technical Report).
- 22. M. Ohta, A. Ikuta and Y. Mitani, "A stochastic signal processing method matched to a feedback function and digitized level observation and its application to an acoustic environment." Transactions of IEE of Japan Vol. 109-C, No. 7, pp. 491-498 (1989) (in Japanese).
- 23. Y. Mitani and M. Ohta, "A practical evaluation method of L eq based on the Beta distribution matched to the restricted fluctuation range in actual random noise," Acustica Vol.69. No.4, pp.185-188 ; (Research Note).
- 24. Y. Mitani, M. Ohta and A. Ikuta, "Time series modeling for stochastic signals by use of generalized adaptive functions and its application to the prediction of acoustic random signals," Proceedings of 1st China-Japan International Symposium on Instrumentation, Neasurement and Automatic Control, pp. 578-585 (1989, Beijing).

REFERENCE PAPERS

- M. Ohta, K. Hatakeyama and Y. Mitani, "A new method of estimating representative statistics of random noise by on-line treatment," Acustica, Vol.51, No. 6, pp. 296-299 (1982); (Research Note).
- M. Ohta, S. Yamaguchi and Y. Mitani, "A new on-line estimation method of representative statistics of environmental random noise and vibration based on actual digital measurement," Journal of the Acoustical Society of Japan Vol. 39, No.6, pp. 361-366 (1983) (in Japanese).
- M. Ohta, S. Yamaguchi and Y. Mitani, "A simplified probabilistic evaluation method for the sound insulation effect of barriers," Bulletin of Australian Acoustical Society Vol. 11, No. 3, pp. 97-103 (1983).
- 4. S. Yamaguchi, M. Ohta, K. Nishimura and Y. Mitani, "A general theory of the state probability for an arbitrary type of nonlinear stochastic energy system based on the additive property of energy and its application to noise environmental system," Proceedings of 1st IASTED International Symposium on Applied Control and Identification Vol.1, pp. 9 -5 - 9-10 (1983, Copenhagen).
- 5. Y. Mitani and M. Ohta, "Probabilistic evaluation method of living environmental noise control system by use of modified statistical energy analysis method," Proceedings of 1983 International Conference on Noise Control Engineering Vol. 2, pp. 1179-1182 (1983, Edinburgh).
- 6. M. Ohta, S. Yamaguchi and Y. Mitani, "A precise noise probability estimation from roughly observed data with quantized levels," Journal of the Acoustical Society of America Vol. 76, No. 1, pp. 122-127 (1984).
- 7. H. Iwashige, M. Ohta, and Y. Mitani, "A probabilistic evaluation of sound energy fluctuations in a room affected by the constant of instrumentation and various absorption materials," Proceedings of 1984 International Conference on Noise Control Engineering Vol. 2, pp. 1209-1212 (1984, Hawaii).
- M. Ohta and Y. Mitani, "A fundamental study for predicting the urban street noise by use of the image method approach," Archives of Acoustics Vol. 10, No. 1, pp. 59-74 (1985).

- 9. M. Ohta and Y. Mitani, "A practical evaluation method for the stochastic noise reduction effect of sound absorbing materials," Acoustics Australia Vol. 14, No. 3, pp. 75-78 (1986).
- 10. M. Ohta and Y. Mitani, "The effect of the restricted fluctuation range in the actual random noise on the transformation of evaluation indices from L<sub>eg</sub> to L<sub>x</sub>." Journal of the Acoustical Society of Japan Vol.43, No.3, pp.143-151 (1987) (in Japanese).
- 11. Y. Mitani and M. Ohta, "A practical probabilistic evaluation method for the noise reduction effect of sound absorbing materials," Journal of the Acoustical Society of Japan (E) Vol.9, No.3, pp.147-152 (1988); (Technical Report).
- 12. M. Ohta and Y. Mitani, "A practical prediction method using an elementary response wave form for the road traffic noise at a T-type intersection," Journal of the Acoustical Society of Japan (E) Vol.9, No.4, pp.195-200 (1988); (Technical Report).
- 13. M. Ohta, A. Ikuta and Y. Mitani, "A stochastic signal processing of analog and digital types for acoustic random power state variable and an establishment of digital filter, "Transactions of the IEICE Vol.E72, No.5, pp.521-530 (1989).
- 14. M. Ohta and Y. Mitani, "Time series prediction method for stochastic acoustic signals by use of an adaptive function model," Measurement Vol.7, No.2, pp. 68-76 (1989).
- 15. M. Ohta, A. Ikuta and Y. Mitani, "A stochastic signal processing method matched to a feedback function and digitized level observation and its application to an acoustic environment," Transactions of IEE of Japan Vol. 109-C, No. 7. pp. 491-498 (1989) (in Japanese).
- 16. Y. Mitani and M. Ohta, "A practical evaluation method of L<sub>eq</sub> based on the Beta distribution matched to the restricted fluctuation range in actual random noise," Acustica Vol.69, No.4, pp.185-188; (Research Note).
- 17. Y. Mitani, M. Ohta and A. Ikuta, "Time series modeling for stochastic signals by use of generalized adaptive functions and its application to the

prediction of acoustic random signals, "Proceedings of 1st China-Japan International Symposium on Instrumentation, Measurement and Automatic Control, pp. 578-585 (1989, Beijing).

A real los recepciones of straining of statistics is large superint concentrational science which is the neural articles is the science of the balance of the science pairs of an animal matrices. In the balance of the science pairs of an animal matrices, we be balance of the science pairs of an animal matrices, we balance of the science of an animal matrices for pairs of the balance of the science of an animal matrices of an animal matrices of the science of an animal matrices of the balance of the science of an animal matrices of the balance of the science of an animal matrices of the balance of the science of the science of an animal matrix and the balance of the science of the science of an animal matrix science of the science of the science of the science of the balance of the science of the science

#### - Theorem and the second states

### i presidente de la companya de la comp

Part State Contesting

Andrea a productive boundary of the second s

and the second s

a the spinished with the set of the set of a set of the set of the

and the second sec

and the second state of the second second

the same of the second se